International Economics

Ninth Edition



Dennis R. Appleyard • Alfred J. Field, Jr.



NINTH EDITION

DENNIS R. APPLEYARD

DAVIDSON COLLEGE

ALFRED J. FIELD, JR.

UNIVERSITY OF NORTH CAROLINA AT CHAPEL HILL





INTERNATIONAL ECONOMICS, NINTH EDITION

Published by McGraw-Hill Education, 2 Penn Plaza, New York, NY 10121. Copyright © 2017 by McGraw-Hill Education. All rights reserved. Printed in the United States of America. Previous editions © 2014, 2010, and 2008. No part of this publication may be reproduced or distributed in any form or by any means, or stored in a database or retrieval system, without the prior written consent of McGraw-Hill Education, including, but not limited to, in any network or other electronic storage or transmission, or broadcast for distance learning.

Some ancillaries, including electronic and print components, may not be available to customers outside the United States.

This book is printed on acid-free paper.

1 2 3 4 5 6 7 8 9 0 DOW 21 20 19 18 17 16

ISBN 978-1-259-29062-6 MHID 1-259-29062-X

Chief Product Officer, SVP Products & Markets: G. Scott Virkler Vice President, General Manager, Products & Markets: Michael Ryan Vice President, Content Design & Delivery: Kimberly Meriwether David Managing Director: James Heine Brand Manager: Katie Hoenicke Director, Product Development: Rose Koos Product Developer: Sarah Otterness Marketing Manager: Virgil Lloyd Director, Content Design & Delivery: Linda Avenarius Program Manager: Mark Christianson Content Project Managers: Jane Mohr, Kristin Bradley, and Karen Jozefowicz Buyer: Jennifer Pickel Design: Studio Montage, St. Louis, MO Content Licensing Specialist: Melisa Seegmiller Cover Image: Ingram Publishing Compositor: SPi Global Printer: R. R. Donnelley

All credits appearing on page or at the end of the book are considered to be an extension of the copyright page.

Library of Congress Cataloging-in-Publication Data

Names: Appleyard, Dennis R., author. | Field, Alfred J., author.
Title: International economics / Dennis R. Appleyard, Davidson College, Alfred J. Field, Jr., University of North Carolina at Chapel Hill.
Description: Ninth edition. | New York, NY : McGraw-Hill Education, [2017]
Identifiers: LCCN 2016016254 | ISBN 9781259290626 (alk. paper)
Subjects: LCSH: International economic relations. | International trade. | International finance.
Classification: LCC HF1359 .A77 2017 | DDC 337—dc23 LC record available at https://lccn.loc.gov/2016016254

The Internet addresses listed in the text were accurate at the time of publication. The inclusion of a website does not indicate an endorsement by the authors or McGraw-Hill Education, and McGraw-Hill Education does not guarantee the accuracy of the information presented at these sites.

The McGraw-Hill Series Economics

ESSENTIALS OF ECONOMICS

Brue, McConnell, and Flynn Essentials of Economics Second Edition

Mandel Economics: The Basics First Edition

Schiller Essentials of Economics Seventh Edition

PRINCIPLES OF ECONOMICS

Colander Economics, Microeconomics, and Macroeconomics Eighth Edition

Frank and Bernanke **Principles of Economics, Principles of Microeconomics, Principles of Macroeconomics** *Fourth Edition*

Frank and Bernanke Brief Editions: Principles of Economics, Principles of Microeconomics, Principles of Macroeconomics First Edition

McConnell, Brue, and Flynn Economics, Microeconomics, and Macroeconomics Eighteenth Edition

McConnell, Brue, and Flynn Brief Editions: Microeconomics and Macroeconomics First Edition

Miller **Principles of Microeconomics** *First Edition*

Samuelson and Nordhaus Economics, Microeconomics, and Macroeconomics Nineteenth Edition

Schiller The Economy Today, The Micro Economy Today, and The Macro Economy Today Twelfth Edition

Slavin Economics, Microeconomics, and Macroeconomics Ninth Edition

ECONOMICS OF SOCIAL ISSUES

Guell Issues in Economics Today Fifth Edition

Sharp, Register, and Grimes Economics of Social Issues Nineteenth Edition

ECONOMETRICS

Gujarati and Porter Basic Econometrics Fifth Edition

Gujarati and Porter Essentials of Econometrics Fourth Edition

MANAGERIAL ECONOMICS

Baye Managerial Economics and Business Strategy Seventh Edition

Brickley, Smith, and Zimmerman Managerial Economics and Organizational Architecture *Fifth Edition*

Thomas and Maurice Managerial Economics Tenth Edition

INTERMEDIATE ECONOMICS

Bernheim and Whinston Microeconomics First Edition

Dornbusch, Fischer, and Startz Macroeconomics Tenth Edition

Frank Microeconomics and Behavior Eighth Edition

ADVANCED ECONOMICS

Romer Advanced Macroeconomics Third Edition

MONEY AND BANKING

Cecchetti Money, Banking, and Financial Markets Second Edition

URBAN ECONOMICS

O'Sullivan **Urban Economics** *Seventh Edition*

LABOR ECONOMICS

Borjas Labor Economics Fifth Edition

McConnell, Brue, and Macpherson Contemporary Labor Economics Ninth Edition

PUBLIC FINANCE

Rosen and Gayer **Public Finance** *Ninth Edition*

Seidman **Public Finance** *First Edition*

ENVIRONMENTAL ECONOMICS

Field and Field Environmental Economics: An Introduction Fifth Edition

INTERNATIONAL ECONOMICS

Appleyard and Field International Economics Ninth Edition

King and King International Economics, Globalization, and Policy: A Reader *Fifth Edition*

Pugel International Economics Fourteenth Edition

ABOUT THE AUTHORS

Dennis R. Appleyard

Dennis R. Appleyard is James B. Duke Professor of International Studies and Professor of Economics, Emeritus, Davidson College, Davidson, North Carolina, and Professor of Economics, Emeritus, University of North Carolina at Chapel Hill. He attended Ohio Wesleyan University for his undergraduate work and the University of Michigan for his Master's and Ph.D. work. He joined the economics faculty at the University of North Carolina at Chapel Hill in 1966 and received the universitywide Tanner Award for "Excellence in Inspirational Teaching of Undergraduate Students" in 1983. He moved to his position at Davidson College in 1990 and retired in 2010. At Davidson, he was Chair of the Department of Economics for seven years and was Director of the college's Semester-in-India Program in fall 2008, and the Semester-in-India and Nepal Program in fall 2000. In 2004 he received Davidson's Thomas Jefferson Award for teaching and service.

Professor Appleyard has taught economic principles, intermediate microeconomics, intermediate macroeconomics, money and banking, international economics, and economic development. His research interests lie in international trade theory and policy and in the Indian economy. Published work, much of it done in conjunction with Professor Field, has appeared in the *American Economic Review, Economic Development and Cultural Change, History of Political Economy, Indian Economic Journal, International Economic Review, Journal of Economic Education,* and *Journal of International Economics*, among others. He has also done consulting work for the World Bank, the U.S. Department of the Treasury, and the Food and Agriculture Organization of the United Nations (in Islamabad, Pakistan). Professor Appleyard always derived genuine pleasure from working with students, and he thinks that teaching kept him young in spirit, since his students were always the same age! He is also firmly convinced that having the opportunity to teach others about international economics in this age of growing globalization is a rare privilege and an enviable challenge.

Alfred J. Field, Jr.

Alfred J. Field is a Professor of Economics, Emeritus, at the University of North Carolina at Chapel Hill. He received his undergraduate and graduate training at Iowa State University and joined the faculty at Carolina in 1967. Field taught courses in international economics and economic development at both the graduate and undergraduate level and directed numerous Senior Honors theses and Master's theses. He served as principal member or director of more than 100 Ph.D. dissertations, duties that he continued to perform after retirement in 2010. In addition, he has served as Director of Graduate Studies, Associate Chair/Director of the Undergraduate Program in Economics, and Acting Department Chair. In 1996, he received the Department's Jae Yeong Song and Chunuk Park Award for Excellence in Graduate Teaching, and in 2006 he received the University of North Carolina at Chapel Hill John L. Sanders Award for Excellence in Undergraduate Teaching and Service. He also served on the Advisory Boards of several university organizations, including the Institute for Latin American Studies.

Professor Field's research encompasses the areas of international trade and economic development. He has worked in Latin America and China, as well as with a number of international agencies in the United States and Europe, primarily on trade and development policy issues. His research interests lie in the areas of trade policy and adjustment and development policy, particularly as they relate to trade, agriculture, and household decision making in developing countries. Another of Field's lines of research addressed trade and structural adjustment issues in the United States, focusing on the textile and apparel industries and the experience of unemployed textile and apparel workers in North Carolina during the 1980s and 1990s. He maintains an active interest in theoretical trade and economic integration issues, as well as the use of econometric and computable general equilibrium models in analyzing the effects of trade policy, particularly in developing countries.

PREFACE

It is our view that in a time of dramatic increase in globalization and high interrelatedness among countries, every student should have a conscious awareness of "things international." Whether one is studying, for example, political science, sociology, chemistry, art, history, or economics, developments worldwide impinge upon the subject matter of the chosen discipline. Such developments may take the form of the discovery of a new compound in Germany, an election result in Greece, a new oil find in Mexico, formation of a new country in Africa, startling new political/terrorist/military events in Pakistan, Syria, France, or the United States, or a change in consumer tastes in China. And, because information now gets transmitted instantaneously across continents and oceans, scientists, governments, firms, and households all react quickly to new information by altering behavior in laboratories, clinics, legislative processes, production and marketing strategies, consumption and travel decisions, and research projects. Without keeping track of international developments, today's student will be unable to understand the changing nature of the world and the material that he or she is studying.

In addition to perceiving the need for international awareness on the part of students in general, we think it is absolutely mandatory that students with an interest in economics recognize that international economic events and the international dimensions of the subject surround us every day. As we prepared to launch this ninth edition of *International Economics*, we could not help noting how much had changed since the initial writing for our first edition. The world has economically internationalized even faster than we anticipated more than 20 years ago, and the awareness of the role of international issues in our lives has increased substantially. Almost daily, headlines focus on developments such as the increased problems facing monetary union in Europe and the euro; proposed policies of erecting additional trade barriers as a protective response to worldwide economic weakness; increased integration efforts such as the emerging Trans-Pacific Partnership; and growing vocal opposition and hostility in many countries to the presence of large and increasing numbers of immigrants at home and abroad. Beyond these broad issues, headlines also trumpet news of the U.S. trade deficit, rising (or falling) gasoline prices, the value of the Chinese renminibi yuan, and the shifting of the headquarters of U.S. firms to foreign locations. In addition, as we write this edition, the world has become painfully aware that increased globalization links countries together strongly in times both of recession and prosperity.

The growing awareness of the importance of international issues is also in evidence in increased student interest in such issues, particularly those related to employment, international working conditions, and equity. It is thus increasingly important that individuals have a practical working knowledge of the economic fundamentals underlying international actions to find their way through the myriad arguments, emotions, and statistics that bombard them almost daily. Young, budding economists need to be equipped with the framework, the tools, and the basic institutional knowledge that will permit them to make sense of the increasingly interdependent economic environment. Further, there will be few jobs that they will later pursue that will not have an international dimension, whether it be ordering components from a Brazilian firm, traveling to a trade show in Malaysia, making a loan for the transport of Caspian Sea oil, or working in an embassy in Quito or in a medical mission in Burundi.

The motive for writing this edition is much the same as in earlier editions: to provide a clear and comprehensive text that will help students move beyond simple recognition and interest in international issues and toward a level of understanding of current and future international developments that will be of use to them in analyzing the problem at hand and selecting a policy position. We seek to help these scholars acquire the necessary human capital for dealing with important questions, for satisfying their intellectual curiosity, and for providing a foundation for future on-the-job decisions.

We have been very flattered by the favorable response to the previous eight editions of our book. In this ninth edition, we continue to build upon the well-received features to develop a text that is current and attuned to our objectives. We have also continued to attempt to clarify our presentation of some of the more difficult concepts and models in order to be more student-friendly.

IMPROVEMENTS AND SPECIFIC CHAPTER CHANGES

In this edition, as usual, we have attempted to provide current and timely information on the wide variety of international economic phenomena. New boxes have been added and previous ones modified to provide up-to-date coverage of emerging issues in the global economy. The text includes such matters as recent developments in U.S. trade policy, major changes in the European Union and developments since the

2007–2009 worldwide financial crisis/recession. We should note that, in the monetary material, we continue to maintain our reliance on the *IS/LM/BP* framework for analyzing macroeconomic policy options, such as during the recent period, because we believe that the framework is effective in facilitating student understanding. We also continue to incorporate key aspects of the asset approach into the *IS/LM/BP* model.

Particular mention should be made of the fact that, in this edition, we have continued to employ Learning Objectives at the beginning of each chapter to orient the reader to the central issues. This text is comprehensive in its coverage of international concepts, and the Learning Objectives are designed to assist the instructor with the choice of chapters to cover in designing the course and to assist the students in focusing on the critical concepts as they begin to read each chapter. The Learning Objectives are now coordinated with the major sections of the chapter in that each objective is tied to a particular section. Further, because of the positive response to the opening vignettes in recent editions, we have retained and updated them in this edition to focus on the real-world applicability of the material.

We have continued to use the pedagogical structure employed in the eighth edition. As in that edition, the "In the Real World" boxes are designed to provide examples of current international issues and developments drawn straight from the news that illustrate the concepts developed in the chapter. We have added, updated, or deleted boxes where appropriate. In situations where particularly critical concepts would benefit from further elaboration or graphical representation, we have continued to utilize "Concept" boxes.

Generally speaking, in each chapter we edited and updated textual material, in addition to the specific changes listed below. Also, where appropriate, we have deleted outdated or overly technical material, and these deletions are not included in this list.

Chapter 1

• Updated all tables and their accompanying descriptions.

Chapter 2

 Introduced additional material to the "In the Real World" box on current thinking with Mercantilist underpinnings and slightly amended existing material in the box.

Chapter 3

 Updated material in the "In the Real World" box on export concentration of selected countries.

Chapter 4

- Extended the table on freight and insurance factors in the "In the Real World" box on transportation costs to include 2013 as well as different countries; introduced new replacement table showing relative transportation costs by major geographical trading area; updated table on containership charter rates.
- Replaced existing Golub study on unit labor costs and trade patterns with newer and more comprehensive examination of the empirical validity of the Ricardian model.
- Introduced recent research on relative productivities and country specialization in output.
- Updated graphs showing U.S. steel industry labor productivity and U.S. import penetration ratios in steel in the existing "In the Real World" box.
- Introduced material on the spread of the benefits of productivity growth in China to other countries.

Chapter 5

• Updated the "In the Real World" box on consumer expenditure patterns since 1960 to include 2014 data.

Chapter 6

 Revised the "In the Real World" box on income distribution and trade to include recent material on changes in welfare due to trade.

Chapter 7

 Updated information contained in "In the Real World" boxes showing the commodity terms of trade and income terms of trade of major groups of countries since 1973.

Chapter 8

- Introduced additional 2014 material on Canadian capital/labor ratios in selected industries.
- Updated estimates of tariff equivalence of transport costs in international trade.
- Restructured and updated the "In the Real World" box on cartels and monopolistic behavior in international trade.

Chapter 9

- Condensed the discussion of the Leontief paradox.
- Added new material on factor-intensity reversals.
- Added new material on the testing of the Heckscher-Ohlin theorem.
- Updated the data on the increasing income inequality in the United States.
- Provided new information on the impact of education on relative wages.

 Added material on the effects of Danish outsourcing.

Chapter 10

- Reorganized the chapter to separate early post– Heckscher-Ohlin theories from more recent post–Heckscher-Ohlin theories.
- Added material on the Melitz model of trade as well as material on multiproduct exporting.

Chapter 11

- Updated table showing the factor endowments of the United States, Japan, Canada, Australia, France, and Mexico to include 2014.
- Revised the "In the Real World" box discussing the recent terms-of-trade movements of four developing countries (Brazil, Jordan, Pakistan, and Thailand).

Chapter 12

- Updated the tables on foreign direct investment abroad by U.S. firms and foreign direct investment in the United States, as well as the tables on the world's largest corporations and the world's largest banks.
- Updated the "In the Real World Box" on migration into the United States and the country origins of that migration.

Chapter 13

• Updated the "In the Real World" box on important features of the commercial policy schedules of Australia, Pakistan, and El Salvador.

Chapter 14

- Added material on the U.S. ban on the export of crude oil.
- Introduced a new "In the Real World" box on the welfare impacts of Japanese import restrictions on rice.
- Introduced material regarding the potential impact of relaxing import restrictions in the context of the Melitz model.
- Added material to the "In the Real World" box on U.S. import restrictions on sugar.

Chapter 15

- Updated the "In the Real World" box on trade taxes as a source of government revenue.
- Updated the "In the Real World" boxes on U.S. antidumping actions and U.S. countervailing duties.
- Added recent developments to the "In the Real World" boxes on Harley-Davidson and Airbus-Boeing.

Chapter 16

 Introduced new Gallup opinion polls on Americans' attitudes toward trade.

- Updated the "In the Real World" box on Trade Adjustment Assistance to include 2015 legislation.
- Updated the discussion of the Doha Development Agenda.
- Did a general re-working of the text's recounting of significant U.S. trade policy actions.

Chapter 17

- Updated the "In the Real World" box on economic integration groups and added the Commonwealth of Independent States.
- Updated material on economic developments in the European Union; in particular, added material on the Greece debt crisis.
- Updated the discussion of NAFTA.
- Added material to the "In the Real World" box on ASEAN.
- Introduced material on the Trans-Pacific Partnership.

Chapter 18

- Changed designation of middle- and lowincome countries from "less developed countries" (LDCs) to "emerging and developing countries" (EDCs).
- Updated the table on economic and noneconomic characteristics of countries at different income levels.
- Updated the table on external debt and debt ratios of emerging and developing countries.
- Re-worked the "In the Real World" box on the Multilateral Debt Relief Initiative (MDRI) to focus on the Heavily Indebted Poor Countries (HIPC) Initiative as well as on the MDRI.

Chapter 19

- Re-worked the material on sample entries in balance-of-payments accounting and on the U.S. balance of payments in light of the IMF's and U.S. Department of Commerce's new format for presenting information on international transactions.
- Updated the table and the discussion of U.S. international transactions.
- Updated the "In the Real World" box on U.S. trade deficits with Japan, China, and Canada; replaced OPEC with Mexico as the 4th economic unit examined.
- Updated the table and discussion of the U.S. international investment position in view of the recent U.S. Department of Commerce change to have direct investment included at its market value rather than at its current cost.
- Updated the "In the Real World" box on the U.S. net international investment position and the U.S. net direct investment position.

Chapter 20

- Updated the "In the Real World" box on the nominal and real Canadian dollar/U.S. dollar exchange rate and the U.S. dollar's nominal and real effective exchange rates.
- Updated the "In the Real World" box on the spot and PPP dollar/euro exchange rates and the spot and PPP dollar/pound exchange rates.
- Updated the Concept Box on currency futures quotations.

Chapter 21

- Updated the table on gross and net international bank lending and the table on the stock of international debt securities.
- Changed the "In the Real World" box on interest rates across countries to a discussion and presentation of (2014) nominal and real lending rates across countries rather than the previous focus on government bond yields across countries.
- Updated the "In the Real World" box on U.S. and eurodollar deposit and lending rates.
- Updated the table and graph on the size of global derivative instruments.

Chapter 22

- Introduced a new opening vignette (on the flow of funds out of the eurozone).
- Updated the discussion and table in the "In the Real World" box dealing with U.S. monetary concepts and the Federal Reserve's balance sheet.
- Introduced recent empirical tests of the monetary approach to the balance of payments and the exchange rate and a test of uncovered interest parity.

Chapter 23

- Added several recent studies to the "In the Real World" box on exchange rate pass-through.
- Expanded the "In the Real World" box on U.S. agricultural exports and the exchange rate to include recent export experience.

Chapter 24

- Updated the "In the Real World" box on average propensities to import of five major countries.
- Introduced a new "In the Real World" box on estimates of the government spending multiplier in developed and developing countries.

Chapter 25

• Updated the "In the Real World" box on foreign exchange restrictions in IMF countries.

Chapter 26

- Updated the "In the Real World" box on the variability of commodity prices and U.S. real GDP.
- Added a brief review of the euro's recent movements to the "In the Real World" box on policy frictions between the European Union and the United States.
- Placed additional emphasis on the G-20 in the "In the Real World" box on macro policy coordination.

Chapter 27

- Updated the "In the Real World" box on actual and natural U.S. levels of GDP and rates of unemployment; replaced previous Robert Gordon estimates with Congressional Budget Office estimates.
- Updated data and discussion in the "In the Real World" box on Sub-Saharan Africa.
- Updated data and discussion in the "In the Real World" box on U.S. inflation and unemployment.

Chapter 28

- Introduced new vignette on problems with both a flexible exchange rate and a fixed exchange rate.
- Updated and included revised information in the "In the Real World" box on reserve holdings under fixed and flexible exchange rates.
- Added brief recent information to the "In the Real World" box on currency boards in Estonia and Lithuania.

Chapter 29

- Introduced new opening vignette pertaining to recent exchange rate movements and the post-Bretton Woods system.
- Introduced a new "In the Real World" box on currency risk and confidence in exchange rates under the 1880–1914 international gold standard.
- Updated the table on country quotas at the IMF.
- Updated the table on the current exchange rate arrangements of countries.
- Updated the table on central bank international reserves, including recognition of the fact that the IMF now values gold at the current London market price rather than at the previous 35 SDRs = 1 ounce of gold price.
- Introduced a new "In the Real World" box on bitcoin.

It is our hope that the changes in the ninth edition will prove beneficial to students as well as to instructors. The improvements are designed to help readers both understand and appreciate more fully the growing importance of the global economy in their lives.

DESCRIPTION OF TEXT

Our book follows the traditional division of international economics into the trade and monetary sides of the subject. Although the primary audience for the book will be students in upper-level economics courses, we think that the material can effectively reach a broad, diversified group of students—including those in political science, international studies, history, and business who may have fewer economics courses in their back-ground. Having taught international economics ourselves in specific nonmajors' sections and Master's of Business Administration sections as well as in the traditional economics department setting, we are confident that the material is accessible to both noneconomics and economics students. This broad audience will be assisted in its learning through the fact that we have included separate, extensive review chapters of microeconomic (Chapter 5) and macroeconomic (Chapter 24) tools.

International Economics presents international trade theory and policy first. Introductory material and data are found in Chapter 1, and Chapters 2 through 4 present the Classical model of trade, including a treatment of pre-Classical Mercantilism. A unique feature is the devotion of an entire chapter to extensions of the Classical model to include more than two countries, more than two goods, money wages and prices, exchange rates, and transportation costs. The analysis is brought forward through the modern Dornbusch-Fischer-Samuelson model including a treatment of the impact of productivity improvements in one country on the trading partner. Chapter 5 provides an extensive review of microeconomic tools used in international trade at this level and can be thought of as a "short course" in intermediate micro. Chapters 6 through 9 present the workhorse neoclassical and Heckscher-Ohlin trade theory, including an examination of the assumptions of the model. Chapter 6 focuses on the traditional production possibilities—indifference curve exposition. We are unabashed fans of the offer curve because of the nice general equilibrium properties of the device and because of its usefulness in analyzing trade policy and in interpreting economic events, and Chapter 7 extensively develops this concept. Chapter 8 explores Heckscher-Ohlin in a theoretical context, and Chapter 9 is unique in its focus on testing the factor endowments approach, including empirical work on the trade-income inequality debate in the context of Heckscher-Ohlin.

Continuing with theory, Chapters 10 through 12 treat extensions of the traditional material. Chapter 10 discusses various post–Heckscher-Ohlin trade theories that relax standard assumptions such as international factor immobility, homogeneous products, constant returns to scale, and perfect competition. An important focus here is upon imperfect competition and intra-industry trade, and new material has been added regarding the multiproduct exporting firm. Chapter 11 explores the comparative statics of economic growth and the relative importance of trade, and it includes material on endogenous growth models and on the effects of growth on the offer curve. Chapter 12 examines causes and consequences of international factor movements, including both capital movements and labor flows.

Chapters 13 through 17 are devoted to trade policy. Chapter 13 is exclusively devoted to presentation of the various instruments of trade policy. Chapter 14 then explores the welfare effects of the instruments, including discussion of such effects in a "small-country" as well as a "large-country" setting. Chapter 15 examines various arguments for protection, including strategic trade policy approaches. Chapter 16 begins with a discussion of the political economy of trade policy, followed by a review of various trade policy actions involving the United States as well as issues currently confronting the WTO. Chapter 17 is a separate chapter on economic integration. We have updated the discussion of the European Union (including recent problems) and the North American Free Trade Agreement. In addition, there is new material on the U.S. free-trade agreements with Colombia, South Korea, and Panama and on the Trans-Pacific Partnership. The trade part of the book concludes with Chapter 18, which provides an overview of how international trade influences growth and change in the emerging/developing countries as well as a discussion of the external debt problem.

The international monetary material begins with Chapter 19, which introduces balance-of-payments accounting. This is followed by discussion of the foreign exchange market in Chapter 20. We think this sequence makes more sense than the reverse, since the demand and supply curves of foreign exchange reflect the debit and credit items, respectively, in the balance of payments. A differentiating feature of the presentation of the foreign exchange market is the extensive development of various exchange rate measures, for example, nominal, real, and effective exchange rates. Chapter 21 then describes characteristics of "real-world" international financial markets in detail, and discusses a (we hope not-too-bewildering) variety of international financial derivative instruments. Chapter 22 presents in considerable detail the monetary and portfolio balance (or asset market) approaches to the balance of payments and to exchange rate determination. The more technical

discussion of testing of these approaches is in an appendix, which has been updated to include recent empirical research. The chapter concludes with an examination of the phenomenon of exchange rate overshooting. In Chapters 23 and 24, our attention turns to the more traditional price and income adjustment mechanisms. Chapter 24 is in effect a review of basic Keynesian macroeconomic analysis.

Chapters 25 through 27 are concerned with macroeconomic policy under different exchange rate regimes. As noted earlier, we continue to utilize the IS/LM/BP Mundell-Fleming approach rather than employ exclusively the asset market approach. The value of the IS/LM/BP model is that it can embrace both the current and the capital/financial accounts in an understandable and perhaps familiar framework for many undergraduates. This model is presented in Chapter 25 in a manner that does not require previous acquaintance with it, but does constitute review material for most students who have previously taken an intermediate macroeconomic theory course. The chapter concludes with an analysis of monetary and fiscal policy in a fixed exchange rate environment. These policies are then examined in a flexible exchange rate environment in Chapter 26. We have included in the appendixes to Chapters 25 and 26 material that develops a more formal graphical link between national income and the exchange rate. The analysis is then broadened to the aggregate demand-aggregate supply framework in Chapter 27. The concluding chapters, Chapters 28 and 29, focus on particular topics of global concern. Chapter 28 considers various issues related to the choice between fixed and flexible exchange rates, including material on currency boards. Chapter 29 then traces the historical development of the international monetary system from Bretton Woods onward, examines proposals for reform such as target zone proposals, and addresses some implications of the 2007-2009 world recession and recent issues with the euro.

Because of the length and comprehensiveness of the *International Economics* text, it is not wise to attempt to cover all of it in a one-semester course. For such a course, we recommend that material be selected from Chapters 1 to 3, 5 to 8, 10, 13 to 15, 19 and 20, 22 to 26, and 29. If more emphasis on international trade is desired, additional material from Chapters 17 and 18 can be included. For more emphasis on international monetary economics, we suggest the addition of selected material from Chapters 21, 27, and 28. For a two-semester course, the entire *International Economics* book can be covered. Whatever the course, occasional outside reading assignments from academic journals, current popular periodicals, a readings book, and Web sources can further help to bring the material to life. The "References for Further Reading" section at the end of the book, which is organized by chapter, can hopefully give some guidance. If library resources are limited, the text contains, both in the main body and in boxes, summaries of some noteworthy contributions.

PEDAGOGICAL DEVICES

To assist the student in learning the material, we have included a variety of pedagogical devices. We like to think of course that the major device in this edition is again clear exposition. Although all authors stress clarity of exposition as a strong point, we continue to be pleased that many reviewers praised this feature. Beyond this general feature, more specific devices are described herein. **Learning Objectives** Except for Chapter 1, every chapter begins with a set of explicit learning objectives to help students focus on key concepts. The learning objectives can also be useful to instructors in selecting material to cover in their respective classes. **Opening Vignettes** These opening vignettes or cases were mentioned earlier. The intent of each case is to motivate the student toward pursuing the material in the forthcoming chapter as well as to enable the student to see how the chapter's topics fit with actual applied situations in the world economy. **Boxes** There are three types of material that appear in boxes (more than 100 of them) in International Economics. Some are analytical in nature (Concept Boxes), and they explain further some difficult concepts or relationships. We have also included several biographical boxes (Titans of International Economics). These short sketches of well-known economists add a personal dimension to the work being studied, and they discuss not only the professional interests and concerns of the individuals but also some of their less well-known "human" characteristics. Finally, the majority of the boxes are case studies (In the Real World), appearing throughout chapters and supplemental to the opening vignettes. These boxes serve to illuminate concepts and analyses under discussion. As with the opening vignettes, they give students an opportunity to see the relevance of the material to current events. They also provide a break from the sometimes heavy dose of theory that permeates international economics texts. **Concept Checks** These are short "stopping points" at various intervals within chapters (about two per chapter). The concept checks pose questions that are designed to see if basic points made in the text have been grasped by the student.

PREFACE

End-of-Chapter Ques-These are standard fare in all texts. The questions and problems are broader and more comprehensive than the questions contained in the concept checks. tions and Problems **Lists of Key Terms** The major terms in each chapter are boldfaced in the chapters themselves and then are brought together at the end of the chapter in list form. A review of each list can serve as a quick review of the chapter. **References for** These lists occur at the end of the book, organized by chapter. We have provided bibliographic sources that we have found useful in our own work as well as entries that are relatively accessible and offer further **Further Reading** theoretical and empirical exploration opportunities for interested students. **Instructor's Manual** This companion work offers instructors assistance in preparing for and teaching the course. We have included suggestions for presenting the material as well as answers to the end-of-chapter questions and and Test Bank problems. In addition, sample examination questions are provided, including some of the hundreds of multiple-choice questions and problems that we have used for examining our own students. Access this ancillary, as well as the Test Bank, through the text's Online Learning Center. **Online Learning** The ninth edition of International Economics is accompanied by a comprehensive website, www.mhhe .com/appleyard/9e. The Instructor's Manual and Test Bank exist in Word format on the password-protected Center portion. Additionally, the password-protected site includes answers to the Graphing Exercises. Students also benefit from visiting the Online Learning Center. Chapter-specific graphing exercises and interactive quizzes serve as helpful study materials. A Digital Image Library contains all of the images from the text. The ninth edition also contains PowerPoint presentations, one to accompany every chapter, available on the Online Learning Center.



CourseSmart is a new way for faculty to find and review eTextbooks. It's also a great option for students who are interested in accessing their course materials digitally. CourseSmart offers thousands of the most commonly adopted textbooks across hundreds of courses from a wide variety of higher education publishers. It is the only place for faculty to review and compare the full text of a textbook online. At CourseSmart, students can save up to 50 percent off the cost of a print book, reduce their impact on the environment, and gain access to powerful Web tools for learning including full text search, notes and highlighting, and e-mail tools for sharing notes between classmates. Complete tech support is also included for each title.

Finding your eBook is easy. Visit www.CourseSmart.com and search by title, author, or ISBN.

ACKNOWLEDGMENTS

Our major intellectual debts are to the many professors who taught us economics, but particularly to Robert Stern of the University of Michigan and Erik Thorbecke of Cornell University. We also have found conversations and seminars over the years with faculty colleagues at the University of North Carolina at Chapel Hill to have been extremely helpful. We particularly wish to thank Stanley Black, Patrick Conway, William A. Darity, Jr., Richard Froyen, and James Ingram. Thanks also to colleagues at Davidson College, especially Peter Hess, Vikram Kumar, David Martin, Lou Ortmayer, and Clark Ross; and to the many students at Chapel Hill and Davidson who were guinea pigs for the material and provided helpful insights and suggestions. In addition, we express special appreciation to Steven L. Cobb of the University of North Texas for his contributions to several previous editions of this book. As a coauthor, Steve provided numerous creative ideas and valuable content, much of which continues to be used in this ninth edition.

We are also indebted to the entire staff at McGraw-Hill, especially Jane Mohr, Sarah Otterness, Katie Hoenicke, and James Heine, as well as freelancer Beth Baugh. We thank them for their cooperation, patience, encouragement, and guidance in the development of this ninth edition. In addition, we are grateful to the following reviewers; their thoughtful, prescriptive comments have helped guide the development of these nine editions:

Deergha Raj Adhikari University of Louisiana at Lafayette

Francis Ahking University of Connecticut–Storrs

Mohsen Bahmani-Oskooee University of Wisconsin–Milwaukee

Scott Baier Clemson University

Michael Barry Mount St. Mary's University

Amitrajeet A. Batabyal Rochester Institute of Technology

Tibor Besedes Georgia Institute of Technology

Bruce Blonigen University of Oregon

Eric Bond Pennsylvania State University

Harry Bowen University of California–Irvine

Josef Brada Arizona State University

Victor Brajer California State University–Fullerton

Charles H. Brayman Kansas State University

Drusilla Brown Tufts University

Geoffrey Carliner Babson College

Roman Cech Longwood University

Winston W. Chang State University of New York at Buffalo

Charles Chittle Bowling Green State University

Patrick Conway University of North Carolina at Chapel Hill

Bienvenido Cortes Pittsburg State University

Kamran Dadkhah Northeastern University

Joseph Daniels Marquette University

William L. Davis University of Tennessee at Martin

Alan Deardorff University of Michigan

Khosrow Doroodian Ohio University-Athens Mary Epps University of Virginia Jim Gerber San Diego State University Norman Gharrity Ohio Wesleyan University Animesh Ghoshal DePaul University William Hallagan Washington State University James Hartigan University of Oklahoma Stephen Haynes University of Oregon Pershing Hill University of Alaska William Hutchinson Vanderbilt University Robert Jerome James Madison University William Kaempfer University of Colorado Mitsuhiro Kaneda Georgetown University Baybars Karacaovali Fordham University Theodore Kariotis Towson University Patrick Kehoe University of Pennsylvania Frank Kelly Indiana University-Purdue University Indianapolis Randall G. Kesselring Arkansas State University David Kemme Wichita State University Madhu Khanna University of Illinois-Champaign Yih-Wu Liu Youngstown State University Thomas Love North Central College Svitlana Maksymenko University of Pittsburgh Judith McDonald Lehigh University

Thomas McGahagan University of Pittsburgh at Johnstown Joseph McKinney Baylor University

Thomas McKinnon University of Arkansas

Michael McPherson University of North Texas

William G. Mertens University of Colorado at Boulder

Thomas Mondschean DePaul University

Michael Moore The George Washington University

Sudesh Mujumdar University of Southern Indiana

Vange Mariet Ocasio University of Denver

John Pomery Purdue University

Michael Quinn Bentley College

James Rakowski University of Notre Dame

James Rauch University of California–San Diego Monica Robayo

University of North Florida

Simran Sahi University of Minnesota Jeff Sarbaum University of North Carolina–Greensboro

W. Charles Sawyer University of Southern Mississippi

Don Schilling University of Missouri

James H. Schindler Columbia Southern University

Modiful Shumon Islam Columbia Southern University

Richard Sicotte University of Vermont

Karen J. Smith Columbia Southern University

John N. Smithin York University

Richard G. Stahl Louisiana State University

Jeffrey Steagall University of North Florida

Grigor Sukiassyan California State University–Fullerton Kishor Thanawala

Villanova University

Edward Tower Duke University

John Wilson Michigan State University

We also wish to thank David Ball (North Carolina State University), David Collie (Cardiff University), David Cushman (University of Saskatchewan), Guzin Erlat (Middle East Technical University–Ankara), J. Michael Finger (World Bank, retired), Dan Friel (Bank of America), Art Goldsmith (Washington and Lee University), the late Monty Graham (The Peterson Institute of International Economics), Michael Jones (Bowdoin College), Joseph Joyce (Wellesley College), Jay Bryson and Nick Bennenbroek (Wells Fargo), and Joe Ross (Goldman Sachs) for their helpful comments on this and earlier editions. Appreciation is also extended to the many other individuals who have contacted us over the years regarding our book. Of course, any remaining shortcomings or errors are the responsibility of the authors (who each blame the other). A special note of thanks goes to our families for their understanding, support, and forbearance throughout the time-absorbing processes required to complete all nine editions.

Finally, we welcome any suggestions or comments that you may have regarding this text. Please feel free to contact us at our e-mail addresses. And thank you for giving attention to our book!

Dennis R. Appleyard deappleyard@davidson.edu Alfred J. Field, Jr. afield@email.unc.edu

BRIEF CONTENTS

CHAPTER 1 The World of International Economics, 1

PART 1

THE CLASSICAL THEORY OF TRADE 15

CHAPTER 2 Early Trade Theories: Mercantilism and the Transition to the Classical World of David Ricardo, 15

CHAPTER 3 The Classical World of David Ricardo and Comparative Advantage, 26

CHAPTER 4 Extensions and Tests of the Classical Model of Trade, 40

PART 2

NEOCLASSICAL TRADE THEORY 62

CHAPTER 5 Introduction to Neoclassical Trade Theory: Tools to Be Employed, 62

CHAPTER 6 Gains from Trade in Neoclassical Theory, 84

CHAPTER 7 Offer Curves and the Terms of Trade, 100

CHAPTER 8 The Basis for Trade: Factor Endowments and the Heckscher-Ohlin Model, 122

CHAPTER 9 Empirical Tests of the Factor Endowments Approach, 150 PART 3

ADDITIONAL THEORIES AND EXTENSIONS 173

CHAPTER 10 Post–Heckscher-Ohlin Theories of Trade and Intra-Industry Trade, 173

CHAPTER 11 Economic Growth and International Trade, 203

CHAPTER 12 International Factor Movements, 226

PART 4

0

TRADE POLICY 257

CHAPTER 13 The Instruments of Trade Policy, 257

CHAPTER 14 The Impact of Trade Policies, 280

CHAPTER 15 Arguments for Interventionist Trade Policies, 320

CHAPTER 16 Political Economy and U.S. Trade Policy, 359

CHAPTER 17 Economic Integration, 387

CHAPTER 18 International Trade and the Developing Countries, 416

PART 5

FUNDAMENTALS OF INTERNATIONAL MONETARY ECONOMICS 448

CHAPTER 19 The Balance-of-Payments Accounts, 448

CHAPTER 20 **The Foreign Exchange Market, 468**

CHAPTER 21 International Financial Markets and Instruments: An Introduction, 498

CHAPTER 22 **The Monetary and Portfolio Balance Approaches to External Balance, 531**

CHAPTER 23 **Price Adjustments and Balance-of-Payments Disequilibrium, 562**

CHAPTER 24 National Income and the Current Account, 590

PART 6

MACROECONOMIC POLICY IN THE OPEN ECONOMY 619

CHAPTER 25 Economic Policy in the Open Economy under Fixed Exchange Rates, 619 CHAPTER 26 Economic Policy in the Open Economy under Flexible Exchange Rates, 651

CHAPTER 27 Prices and Output in the Open Economy: Aggregate Supply and Demand, 673

PART 7

ISSUES IN WORLD MONETARY ARRANGEMENTS 701

CHAPTER 28 Fixed or Flexible Exchange Rates? 701

CHAPTER 29 The International Monetary System: Past, Present, and Future, 728

References for Further Reading, 765

Index, 784

CONTENTS

CHAPTER 1

The World of International Economics, 1

INTRODUCTION, 1
THE NATURE OF MERCHANDISE TRADE, 3
The Geographical Composition of Trade, 3
The Commodity Composition of Trade, 6
U.S. International Trade, 7
WORLD TRADE IN SERVICES, 9
THE CHANGING DEGREE OF ECONOMIC INTERDEPENDENCE, 11
SUMMARY, 12
Appendix, A General Reference List in International Economics, 12

PART 1

THE CLASSICAL THEORY OF TRADE 15

CHAPTER 2

Early Trade Theories: Mercantilism and the Transition to the **Classical World of David Ricardo**, 15 INTRODUCTION, 16 The Oracle in the 21st Century, 16 MERCANTILISM, 16 The Mercantilist Economic System, 16 The Role of Government, 17 Mercantilism and Domestic Economic Policy, 18 IN THE REAL WORLD: MERCANTILISM IS STILL ALIVE 19 THE CHALLENGE TO MERCANTILISM BY EARLY CLASSICAL WRITERS, 20 David Hume-The Price-Specie-Flow Mechanism, 20 CONCEPT BOX 1: CAPSULE SUMMARY OF THE PRICE-Specie-Flow Mechanism, 20 CONCEPT BOX 2: CONCEPT REVIEW—PRICE ELASTICITY AND TOTAL EXPENDITURES, 21 Adam Smith and the Invisible Hand, 22 TITANS OF INTERNATIONAL ECONOMICS: Adam Smith (1723–1790), 23 SUMMARY, 24

CHAPTER 3

The Classical World of David Ricardo and

Comparative Advantage, 26

INTRODUCTION, 27
Some Common Myths, 27
ASSUMPTIONS OF THE BASIC RICARDIAN MODEL, 27 *TITANS OF INTERNATIONAL ECONOMICS: David Ricardo* (1772–1823), 28
RICARDIAN COMPARATIVE ADVANTAGE, 28 *IN THE REAL WORLD: Export Concentration of Selected* COUNTRIES, 31
COMPARATIVE ADVANTAGE AND THE TOTAL GAINS FROM TRADE, 32
Resource Constraints, 32
Complete Specialization, 33

REPRESENTING THE RICARDIAN MODEL WITH PRODUCTION-POSSIBILITIES FRONTIERS, 34 Production Possibilities—An Example, 34 Maximum Gains from Trade, 36 COMPARATIVE ADVANTAGE—SOME CONCLUDING OBSERVATIONS, 37 SUMMARY, 38

CHAPTER 4

Extensions and Tests of the Classical Model of Trade, 40

INTRODUCTION, 41 Trade Complexities in the Real World, 41 THE CLASSICAL MODEL IN MONEY TERMS, 41 WAGE RATE LIMITS AND EXCHANGE RATE LIMITS, 42 CONCEPT BOX 1: WAGE RATE LIMITS AND EXCHANGE RATE LIMITS IN THE MONETIZED RICARDIAN FRAMEWORK, 44 **MULTIPLE COMMODITIES**, 45 The Effect of Wage Rate Changes, 46 The Effect of Exchange Rate Changes, 47 **TRANSPORTATION COSTS**, 48 IN THE REAL WORLD: THE SIZE OF TRANSPORTATION COSTS, 49 **MULTIPLE COUNTRIES, 50 EVALUATING THE CLASSICAL MODEL**, 51 IN THE REAL WORLD: LABOR PRODUCTIVITY AND IMPORT PENETRATION IN THE U.S. STEEL INDUSTRY, 54 IN THE REAL WORLD: EXPORTING AND PRODUCTIVITY, 56 SUMMARY, 56 Appendix, The Dornbusch, Fischer, and Samuelson Model, 58

PART 2

NEOCLASSICAL TRADE THEORY 62

CHAPTER 5 **Introduction to Neoclassical Trade Theory: Tools to Be Employed. 62** INTRODUCTION, 63 THE THEORY OF CONSUMER BEHAVIOR, 63 Consumer Indifference Curves, 63 TITANS OF INTERNATIONAL ECONOMICS: FRANCIS YSIDRO EDGEWORTH (1845–1926), 64 The Budget Constraint, 68 Consumer Equilibrium, 69 **PRODUCTION THEORY**, 70 Isoquants, 70 IN THE REAL WORLD: CONSUMER EXPENDITURE PATTERNS IN THE UNITED STATES, 71 Isocost Lines, 73 Producer Equilibrium, 75 THE EDGEWORTH BOX DIAGRAM AND THE PRODUCTION-Possibilities Frontier, 75

The Edgeworth Box Diagram, 75 The Production-Possibilities Frontier, 78 SUMMARY, 82

CHAPTER 6

Gains from Trade in Neoclassical Theory, 84 INTRODUCTION, 85 The Effects of Restrictions on U.S. Trade, 85 AUTARKY EOUILIBRIUM, 85 INTRODUCTION OF INTERNATIONAL TRADE, 87 The Consumption and Production Gains from Trade, 89 Trade in the Partner Country, 91 MINIMUM CONDITIONS FOR TRADE, 92 Trade between Countries with Identical PPFs, 92 Trade between Countries with Identical Demand Conditions, 93 Conclusions, 95 SOME IMPORTANT ASSUMPTIONS IN THE ANALYSIS, 95 Costless Factor Mobility, 95 Full Employment of Factors of Production, 95 The Indifference Curve Map Can Show Welfare Changes, 96 IN THE REAL WORLD: CHANGES IN INCOME DISTRIBUTION AND WELFARE WITH INCREASED TRADE, 97 SUMMARY, 98 Appendix, "Actual" versus "Potential" Gains from Trade, 99

CHAPTER 7

Offer Curves and the Terms of Trade, 100 INTRODUCTION, 101 Terms-of-Trade Shocks, 101 A COUNTRY'S OFFER CURVE, 101 CONCEPT BOX 1: THE TABULAR APPROACH TO DERIVING AN OFFER CURVE, 104 TRADING EQUILIBRIUM, 105 SHIFTS OF OFFER CURVES, 107 CONCEPT BOX 2: MEASUREMENT OF THE TERMS OF TRADE, 110 ELASTICITY AND THE OFFER CURVE, 111 IN THE REAL WORLD: TERMS OF TRADE FOR MAJOR GROUPS OF COUNTRIES, 1973-2013, 112 OTHER CONCEPTS OF THE TERMS OF TRADE, 116 Income Terms of Trade, 116 Single Factoral Terms of Trade, 116 IN THE REAL WORLD: INCOME TERMS OF TRADE OF MAJOR GROUPS OF COUNTRIES, 1973–2013, 117 Double Factoral Terms of Trade, 118 SUMMARY, 118 Appendix A, Derivation of Import-Demand Elasticity on an Offer Curve, 119 Appendix B, Elasticity and Instability of Offer Curve Equilibria, 120

CHAPTER 8

The Basis for Trade: Factor Endowments and the Heckscher-Ohlin Model, 122 INTRODUCTION, 123

Do Labor Standards Affect Comparative Advantage? 123

FACTOR ENDOWMENTS AND THE HECKSCHER-**Ohlin Theorem**, 124 Factor Abundance and Heckscher-Ohlin, 124 Commodity Factor Intensity and Heckscher-Ohlin, 125 IN THE REAL WORLD: RELATIVE FACTOR ENDOWMENTS IN Selected Countries, 126 The Heckscher-Ohlin Theorem, 128 IN THE REAL WORLD: RELATIVE FACTOR INTENSITIES IN CANADA, 128 TITANS OF INTERNATIONAL ECONOMICS: PAUL ANTHONY SAMUELSON (1915–2009), 132 TRADE, FACTOR PRICES, AND INCOME DISTRIBUTION 132 The Factor Price Equalization Theorem, 132 The Stolper-Samuelson Theorem and Income Distribution Effects of Trade in the Heckscher-Ohlin Model, 135 Conclusions, 137 THEORETICAL QUALIFICATIONS TO HECKSCHER-Ohlin, 137 Demand Reversal, 137 Factor-Intensity Reversal, 138 Transportation Costs, 139 Imperfect Competition, 141 Immobile or Commodity-Specific Factors, 142 IN THE REAL WORLD: MONOPOLY BEHAVIOR IN INTERNATIONAL TRADE, 144 Other Considerations, 147 CONCEPT BOX 1: THE SPECIFIC-FACTORS MODEL AND THE Real Wage of Workers, 147 SUMMARY, 149

CHAPTER 9

Empirical Tests of the Factor Endowments Approach, 150 INTRODUCTION, 151 Theories, Assumptions, and the Role of Empirical Work, 151 THE LEONTIEF PARADOX, 151 INITIAL EXPLANATIONS FOR THE LEONTIEF PARADOX, 152 Demand Reversal, 152 Factor-Intensity Reversal, 153 IN THE REAL WORLD: TESTING FOR FACTOR-INTENSITY REVERSALS, 154 U.S. Tariff Structure, 155 Different Skill Levels of Labor, 155 The Role of Natural Resources, 156 MORE RECENT TESTS OF THE HECKSCHER-OHLIN THEOREM, 157 Factor Content Approach with Many Factors, 158 Technology, Productivity, and "Home Bias", 161 IN THE REAL WORLD: HECKSCHER-OHLIN AND COMPARATIVE Advantage, 163 HECKSCHER-OHLIN AND INCOME INEQUALITY, 164 IN THE REAL WORLD: TRADE AND INCOME INEQUALITY IN A LESS DEVELOPED COUNTRY: THE CASE OF MOZAMBIQUE, 168 IN THE REAL WORLD: OUTSOURCING AND WAGE INEQUALITY, 170 SUMMARY, 172

PART 3

ADDITIONAL THEORIES AND EXTENSIONS 173

CHAPTER 10

Post–Heckscher-Ohlin Theories of Trade and Intra-Industry Trade, 173

INTRODUCTION, 174 A Trade Myth, 174 EARLY POST-HECKSCHER-OHLIN THEORIES OF TRADE, 174 The Imitation Lag Hypothesis, 174 The Product Cycle Theory, 175 The Linder Theory, 178 Economies of Scale, 181 MORE RECENT ALTERNATIVE TRADE THEORIES, 182 The Krugman Model, 182 The Reciprocal Dumping Model, 185 Vertical Specialization-Based Trade, 186 IN THE REAL WORLD: New Venture INTERNATIONALIZATION, 187 Firm-Focused Theories, 188 The Gravity Model, 188 The Melitz Model and Multiproduct Exporting, 189 Concluding Comments on Post-Heckscher-Ohlin Trade Theories, 191 IN THE REAL WORLD: GEOGRAPHY AND TRADE, 192 INTRA-INDUSTRY TRADE, 193 Reasons for Intra-Industry Trade in a Product Category, 194 The Level of a Country's Intra-Industry Trade, 196 SUMMARY, 197 Appendix A, Economies of Scale, 198 Appendix B, Monopolistic Competition and Price Elasticity of Demand in the Krugman Model, 200 Appendix C, Measurement of Intra-Industry Trade, 202

CHAPTER 11

Economic Growth and International Trade, 203 INTRODUCTION, 204 China—A Regional Growth Pole, 204 CLASSIFYING THE TRADE EFFECTS OF ECONOMIC GROWTH, 204 Trade Effects of Production Growth, 204 Trade Effects of Consumption Growth, 206 SOURCES OF GROWTH AND THE PRODUCTION-POSSIBILITIES FRONTIER, 208 The Effects of Technological Change, 208 IN THE REAL WORLD: LABOR AND CAPITAL REQUIREMENTS PER UNIT OF OUTPUT, 209 IN THE REAL WORLD: "SPILLOVERS" AS A CONTRIBUTOR TO ECONOMIC GROWTH, 212

The Effects of Factor Growth, 212

Factor Growth, Trade, and Welfare in the Small-Country Case, 215

GROWTH, TRADE, AND WELFARE: THE LARGE-COUNTRY CASE, 216

CONCEPT BOX 1: LABOR FORCE GROWTH AND PER CAPITA INCOME, 217
CONCEPT BOX 2: ECONOMIC GROWTH AND THE OFFER CURVE, 219
GROWTH AND THE TERMS OF TRADE: A DEVELOPING-COUNTRY PERSPECTIVE, 221
IN THE REAL WORLD: TERMS OF TRADE OF BRAZIL, JORDAN, PAKISTAN, AND THAILAND, 1980–2014, 222
SUMMARY, 224

CHAPTER 12

International Factor Movements, 226 INTRODUCTION, 227

INTERNATIONAL CAPITAL MOVEMENTS THROUGH FOREIGN DIRECT INVESTMENT AND MULTINATIONAL CORPORATIONS, 227 Foreign Investors in China: "Good" or "Bad" from the Chinese Perspective? 227 Definitions, 229 Some Data on Foreign Direct Investment and Multinational Corporations, 229 Reasons for International Movement of Capital, 232 IN THE REAL WORLD: DETERMINANTS OF FOREIGN DIRECT INVESTMENT, 234 THE EFFECTS OF INTERNATIONAL CAPITAL MOVEMENTS 236 A Theoretical Framework for Analyzing International Capital Movements, 236 IN THE REAL WORLD: HOST-COUNTRY DETERMINANTS OF Foreign Direct Investment Inflows, 237 Potential Benefits and Costs of Foreign Direct Investment to a Host Country, 239 LABOR MOVEMENTS BETWEEN COUNTRIES, 242 Seasonal Workers in Germany, 242 Permanent Migration: A Greek in Germany, 243 IN THE REAL WORLD: MIGRATION FLOWS INTO THE UNITED STATES, 1986 AND 2013, 244 Economic Effects of Labor Movements, 245 Additional Considerations Pertaining to International Migration, 248 IN THE REAL WORLD: IMMIGRANT REMITTANCES, 249 Immigration and the United States-Recent Perspectives, 252 IN THE REAL WORLD: IMMIGRATION AND TRADE, 253 IN THE REAL WORLD: IMMIGRATION INTO THE UNITED STATES

and the Brain Drain from Developing Countries, 255 Summary, 256

PART 4

TRADE POLICY 257

CHAPTER 13

The Instruments of Trade Policy, 257

INTRODUCTION, 258

In What Ways Can Governments Interfere with Trade?, 258 IMPORT TARIFFS, 259

Specific Tariffs, 259 Ad Valorem Tariffs, 259 Other Features of Tariff Schedules, 259 IN THE REAL WORLD: U.S. TARIFF RATES, 261 IN THE REAL WORLD: THE U.S. GENERALIZED SYSTEM OF PREFERENCES, 263 Measurement of Tariffs, 265 IN THE REAL WORLD: NOMINAL AND EFFECTIVE TARIFFS IN THE EUROPEAN UNION, 267 IN THE REAL WORLD: NOMINAL AND EFFECTIVE TARIFF RATES IN VIETNAM AND EGYPT, 268 EXPORT TAXES AND SUBSIDIES, 270 NONTARIFF BARRIERS TO FREE TRADE, 271 Import Quotas, 271 "Voluntary" Export Restraints (VERs), 271 Government Procurement Provisions, 271 Domestic Content Provisions, 272 European Border Taxes, 272 Administrative Classification, 273 Restrictions on Services Trade, 273 Trade-Related Investment Measures, 273 Additional Restrictions, 273 Additional Domestic Policies That Affect Trade, 274 IN THE REAL WORLD: IS IT A CAR? IS IT A TRUCK? 274 IN THE REAL WORLD: EXAMPLES OF CONTROL OVER TRADE, 275 IN THE REAL WORLD: THE EFFECT OF PROTECTION INSTRUMENTS ON DOMESTIC PRICES, 276 SUMMARY, 278

CHAPTER 14

The Impact of Trade Policies, 280 INTRODUCTION, 281 Gainers and Losers from Steel Tariffs, 281 TRADE RESTRICTIONS IN A PARTIAL EQUILIBRIUM SETTING: THE **SMALL-COUNTRY CASE**, 282 The Impact of an Import Tariff, 282 The Impact of an Import Quota and a Subsidy to Import-Competing Production, 285 The Impact of Export Policies, 288 IN THE REAL WORLD: REAL INCOME GAINS FROM TRADE LIBERALIZATION IN AGRICULTURE, 289 TRADE RESTRICTIONS IN A PARTIAL EQUILIBRIUM SETTING: THE LARGE-COUNTRY CASE, 291 Framework for Analysis, 291 The Impact of an Import Tariff, 294 The Impact of an Import Quota, 297 The Impact of an Export Tax, 299 IN THE REAL WORLD: WELFARE COSTS OF U.S. IMPORT QUOTAS AND VERS, 301 The Impact of an Export Subsidy, 302 TRADE RESTRICTIONS IN A GENERAL EQUILIBRIUM SETTING, 303 Protection in the Small-Country Case, 303 IN THE REAL WORLD: WELFARE EFFECTS OF RESTRICTIVE RICE POLICIES IN JAPAN, 304 Protection in the Large-Country Case, 306 OTHER EFFECTS OF PROTECTION, 309

IN THE REAL WORLD: DOMESTIC EFFECTS OF THE SUGAR QUOTA SYSTEM, 310
SUMMARY, 312
Appendix A, The Impact of Protection in a Market with Nonhomogeneous Goods, 313
Appendix B, The Impact of Trade Policy in the Large-Country Setting Using Export Supply and Import Demand Curves, 314

CHAPTER 15 **Arguments for Interventionist Trade Policies, 320** INTRODUCTION, 321 Calls for Protection, 321 TRADE POLICY AS A PART OF BROADER SOCIAL POLICY **OBJECTIVES FOR A NATION, 321** Trade Taxes as a Source of Government Revenue, 322 National Defense Argument for a Tariff, 322 IN THE REAL WORLD: THE RELATIVE IMPORTANCE OF TRADE TAXES AS A SOURCE OF GOVERNMENT REVENUE, 323 Tariff to Improve the Balance of Trade, 324 The Terms-of-Trade Argument for Protection, 325 Tariff to Reduce Aggregate Unemployment, 327 Tariff to Increase Employment in a Particular Industry, 328 IN THE REAL WORLD: INDUSTRY EMPLOYMENT EFFECTS OF TRADE LIBERALIZATION, 328 IN THE REAL WORLD: COSTS OF PROTECTING INDUSTRY Employment, 329 Tariff to Benefit a Scarce Factor of Production, 329 Fostering "National Pride" in Key Industries, 330 Differential Protection as a Component of a Foreign Policy/ Aid Package, 330 PROTECTION TO OFFSET MARKET IMPERFECTIONS, 331 The Presence of Externalities as an Argument for Protection, 331 Tariff to Extract Foreign Monopoly Profit, 333 The Use of an Export Tax to Redistribute Profit from a Domestic Monopolist, 334 PROTECTION AS A RESPONSE TO INTERNATIONAL POLICY DISTORTIONS, 335 Tariff to Offset Foreign Dumping, 335 Tariff to Offset a Foreign Subsidy, 336 IN THE REAL WORLD: ANTIDUMPING ACTIONS IN THE UNITED STATES 337 IN THE REAL WORLD: COUNTERVAILING DUTIES IN THE UNITED STATES, 339 MISCELLANEOUS, INVALID ARGUMENTS, 341 STRATEGIC TRADE POLICY: FOSTERING COMPARATIVE Advantage, 341 The Infant Industry Argument for Protection, 342 IN THE REAL WORLD: U.S. MOTORCYCLES—A SUCCESSFUL INFANT INDUSTRY? 343 Economies of Scale in a Duopoly Framework, 344 Research and Development and Sales of a Home Firm, 347 Export Subsidy in Duopoly, 349 Strategic Government Interaction and World Welfare, 352 IN THE REAL WORLD: AIRBUS AND BOEING, 353 Concluding Observations on Strategic Trade Policy, 355 SUMMARY, 356

CHAPTER 16 Political Economy and U.S. Trade Policy, 359 INTRODUCTION, 360 Contrasting Vignettes on Trade Policy, 360 THE POLITICAL ECONOMY OF TRADE POLICY, 360 The Self-Interest Approach to Trade Policy, 361 IN THE REAL WORLD: WORLD ATTITUDES TOWARD FOREIGN TRADE, 362 IN THE REAL WORLD: U.S. ATTITUDES TOWARD INTERNATIONAL TRADE, 363 The Social Objectives Approach, 364 IN THE REAL WORLD: POLITICS PUTS THE SQUEEZE ON TOMATO IMPORTS, 365 A REVIEW OF U.S. TRADE POLICY AND MULTILATERAL NEGOTIATIONS, 366 Reciprocal Trade Agreements and Early GATT Rounds, 367 The Kennedy Round of Trade Negotiations, 367 The Tokyo Round of Trade Negotiations, 368 IN THE REAL WORLD: TRADE ADJUSTMENT ASSISTANCE AND ITS IMPLEMENTATION, 370 The Uruguay Round of Trade Negotiations, 371 Trade Policy Issues after the Uruguay Round, 373 IN THE REAL WORLD: TARIFF REDUCTIONS RESULTING FROM THE URUGUAY ROUND, 374 IN THE REAL WORLD: NATIONAL SOVEREIGNTY AND THE WORLD TRADE ORGANIZATION, 377 The Doha Development Agenda, 377 Recent U.S. Actions, 380 **CONCLUDING OBSERVATIONS ON TRADE POLICY, 384** The Conduct of Trade Policy, 384 SUMMARY, 385 CHAPTER 17 **Economic Integration**, 387

INTRODUCTION, 388 Promise and Problems of Integration, 388 **TYPES OF ECONOMIC INTEGRATION, 388** Free-Trade Area, 388 Customs Union, 389 Common Market, 389 Economic Union, 389 THE STATIC AND DYNAMIC EFFECTS OF ECONOMIC INTEGRATION, 389 Static Effects of Economic Integration, 389 IN THE REAL WORLD: ECONOMIC INTEGRATION UNITS, 391 IN THE REAL WORLD: TRADE CREATION AND TRADE DIVERSION IN THE EARLY STAGES OF EUROPEAN ECONOMIC INTEGRATION, 392 General Conclusions on Trade Creation/Diversion, 396 CONCEPT BOX 1: TRADE DIVERSION IN GENERAL Equilibrium, 396 Dynamic Effects of Economic Integration, 398 Summary of Economic Integration, 398 THE EUROPEAN UNION, 399 History and Structure, 399 IN THE REAL WORLD: THE EAST AFRICAN COMMUNITY, 400

Early Growth and Disappointments, 401 Completing the Internal Market, 401 U.S. ECONOMIC INTEGRATION AGREEMENTS, 403 **NAFTA**, 403 Effects of NAFTA, 405 IN THE REAL WORLD: NAFTA-MYTHS VS. FACTS, 408 Recent U.S. Integration Agreements, 409 OTHER MAJOR ECONOMIC INTEGRATION EFFORTS, 411 MERCOSUR, 411 FTAA, 411 Chilean Trade Agreements, 412 APEC, 412 Trans-Pacific Partnership, 412 IN THE REAL WORLD: ASIAN ECONOMIC INTERDEPENDENCE LEADS TO GREATER INTEGRATION, 413 Transatlantic Trade and Investment Partnership, 414 SUMMARY, 414

CHAPTER 18

International Trade and the Developing Countries, 416 INTRODUCTION, 417 Recovery in East Asia after Financial Crisis, 417 AN OVERVIEW OF THE DEVELOPING COUNTRIES, 417 THE ROLE OF TRADE IN FOSTERING ECONOMIC Development, 418 The Static Effects of Trade on Economic Development, 419 The Dynamic Effects of Trade on Development, 420 Export Instability, 421 Potential Causes of Export Instability, 422 Long-Run Terms-of-Trade Deterioration, 423 TITANS OF INTERNATIONAL ECONOMICS: RAUL PREBISCH (1901–1986) AND HANS WOLFGANG SINGER (1910-2006), 425 Empirical Evidence on Trade and Development, 427 TRADE POLICY AND THE DEVELOPING COUNTRIES, 429 Policies to Stabilize Export Prices or Earnings, 429 Problems with International Commodity Agreements, 430 Suggested Policies to Combat a Long-Run Deterioration in the Terms of Trade, 430 IN THE REAL WORLD: MANAGING PRICE INSTABILITY, 431 IN THE REAL WORLD: THE LENGTH OF COMMODITY PRICE SHOCKS, 431 Inward-Looking versus Outward-Looking Trade Strategies, 434 IN THE REAL WORLD: Emerging Connections between ASIA AND AFRICA, 437 THE EXTERNAL DEBT PROBLEM OF THE DEVELOPING COUNTRIES, 438 Causes of the Developing Countries' Debt Problem, 439 Possible Solutions to the Debt Problem, 440 IN THE REAL WORLD: THE HIPC AND MDRI DEBT RELIEF INITIATIVES, 443 SUMMARY, 446

PART 5

FUNDAMENTALS OF INTERNATIONAL MONETARY ECONOMICS 448

CHAPTER 19

The Balance-of-Payments Accounts, 448

INTRODUCTION, 449 China's Trade Surpluses and Deficits, 449 CREDITS, DEBITS, AND SAMPLE ENTRIES IN BALANCE-OF-PAYMENTS ACCOUNTING, 450 ASSEMBLING A BALANCE-OF-PAYMENTS SUMMARY STATEMENT, 453 IN THE REAL WORLD: CURRENT ACCOUNT DEFICITS, 457 BALANCE-OF-PAYMENTS SUMMARY STATEMENT FOR THE UNITED STATES, 459 IN THE REAL WORLD: U.S. TRADE DEFICITS WITH CANADA, CHINA, JAPAN, AND MEXICO, 460 INTERNATIONAL INVESTMENT POSITION OF THE UNITED STATES, 463 IN THE REAL WORLD: TRENDS IN THE U.S. INTERNATIONAL **INVESTMENT POSITION**, 465 SUMMARY, 466

CHAPTER 20

The Foreign Exchange Market, 468

INTRODUCTION, 469 The Yen Also Rises (and Falls), 469 THE FOREIGN EXCHANGE RATE AND THE MARKET FOR FOREIGN EXCHANGE, 469 Demand Side, 469 Supply Side, 470 The Market, 470 THE SPOT MARKET, 472 Principal Actors, 473 The Role of Arbitrage, 473 Different Measures of the Spot Rate, 474 IN THE REAL WORLD: NOMINAL AND REAL EXCHANGE RATES OF THE U.S. DOLLAR, 477 THE FORWARD MARKET, 480 IN THE REAL WORLD: SPOT AND PPP EXCHANGE RATES, 482 CONCEPT BOX 1: CURRENCY FUTURES QUOTATIONS, 486 THE LINK BETWEEN THE FOREIGN EXCHANGE MARKETS AND THE FINANCIAL MARKETS, 487 The Basis for International Financial Flows, 487 Covered Interest Parity and Financial Market Equilibrium, 490 Simultaneous Adjustment of the Foreign Exchange Markets and the Financial Markets, 493 SUMMARY, 496

CHAPTER 21

International Financial Markets and Instruments: An Introduction, 498 INTRODUCTION, 499

Financial Globalization: A Recent Phenomenon? 499

INTERNATIONAL BANK LENDING, 499 THE INTERNATIONAL FINANCIAL MARKETS, 505 International Bond Market (Debt Securities) 505 IN THE REAL WORLD: INTEREST RATES ACROSS COUNTRIES. 508 International Stock Markets, 510 FINANCIAL LINKAGES AND EUROCURRENCY DERIVATIVES, 512 Basic International Financial Linkages: A Review, 513 International Financial Linkages and the Eurodollar Market, 514 IN THE REAL WORLD: U.S. DOMESTIC AND EURODOLLAR Deposit and Lending Rates, 1989-2014, 516 Hedging Eurodollar Interest Rate Risk, 518 CONCEPT BOX 1: EURODOLLAR INTEREST RATE FUTURES MARKET OUOTATIONS, 523 THE CURRENT GLOBAL DERIVATIVES MARKET, 527 SUMMARY, 529

CHAPTER 22

The Monetary and Portfolio Balance Approaches to **External Balance**, 531 INTRODUCTION, 532 International Interdependence of Money, 532 THE MONETARY APPROACH TO THE BALANCE OF PAYMENTS, 532 The Supply of Money, 533 The Demand for Money, 534 IN THE REAL WORLD: RELATIONSHIPS BETWEEN MONETARY CONCEPTS IN THE UNITED STATES, 535 Monetary Equilibrium and the Balance of Payments, 537 THE MONETARY APPROACH TO THE EXCHANGE RATE, 539 A Two-Country Framework, 540 THE PORTFOLIO BALANCE APPROACH TO THE BALANCE OF PAYMENTS AND THE EXCHANGE RATE, 542 Asset Demands, 542 Portfolio Balance, 544 Portfolio Adjustments, 545 EXCHANGE RATE OVERSHOOTING, 548 TITANS OF INTERNATIONAL ECONOMICS: RUDIGER DORNBUSCH (1942-2002), 549 SUMMARY, 555 Appendix, Examples of Empirical Work on the Monetary and Portfolio Balance Approaches, 556

CHAPTER 23

Price Adjustments and Balance-of-Payments Disequilibrium, 562 INTRODUCTION, 563 Price Adjustment: The Exchange Rate Question, 563 THE PRICE ADJUSTMENT PROCESS AND THE CURRENT ACCOUNT UNDER A FLEXIBLE-RATE SYSTEM, 563 The Demand for Foreign Goods and Services and the Foreign Exchange Market, 564 Market Stability and the Price Adjustment Mechanism, 567 CONCEPT BOX 1: ELASTICITY OF IMPORT DEMAND AND THE SUPPLY CURVE OF FOREIGN EXCHANGE WHEN DEMAND IS LINEAR, 571

The Price Adjustment Process: Short Run versus Long Run, 574

- IN THE REAL WORLD: ESTIMATES OF IMPORT AND EXPORT DEMAND ELASTICITIES, 575
- IN THE REAL WORLD: ESTIMATES OF EXCHANGE RATE PASS-THROUGH, 577
- IN THE REAL WORLD: JAPANESE EXPORT PRICING AND PASS-THROUGH IN THE 1990s, 578
- IN THE REAL WORLD: U.S. AGRICULTURAL EXPORTS AND EXCHANGE RATE CHANGES, 582

THE PRICE ADJUSTMENT MECHANISM IN A FIXED EXCHANGE RATE SYSTEM, 583 Gold Standard, 583 The Price Adjustment Mechanism and the Pegged Rate System, 585 SUMMARY, 586

Appendix, Derivation of the Marshall-Lerner Condition, 588

CHAPTER 24

National Income and the Current Account, 590 INTRODUCTION, 591 Does GDP Growth Cause Trade Deficits? 591 THE CURRENT ACCOUNT AND NATIONAL INCOME, 591 The Keynesian Income Model, 591 TITANS OF INTERNATIONAL ECONOMICS: JOHN MAYNARD Keynes (1883–1946), 592 Determining the Equilibrium Level of National Income, 597 IN THE REAL WORLD: AVERAGE PROPENSITIES TO IMPORT, Selected Countries, 598 THE MULTIPLIER PROCESS, 603 The Autonomous Spending Multiplier, 603 IN THE REAL WORLD: MULTIPLIER ESTIMATES FOR INDIA, 605 The Current Account and the Multiplier, 606 IN THE REAL WORLD: THE GOVERNMENT SPENDING MULTIPLIER IN DEVELOPED AND DEVELOPING COUNTRIES, 607 Foreign Repercussions and the Multiplier Process, 608 IN THE REAL WORLD: HISTORICAL CORRELATION OVER TIME OF COUNTRIES' GDP, 610 AN OVERVIEW OF PRICE AND INCOME ADJUSTMENTS AND SIMULTANEOUS EXTERNAL AND INTERNAL BALANCE, 610 IN THE REAL WORLD: RECENT SYNCHRONIZATION OF GDP MOVEMENTS OF COUNTRIES, 611 SUMMARY, 613 Appendix A, The Multiplier When Taxes Depend on Income, 614 Appendix B, Derivation of the Multiplier with Foreign

Repercussions, 616

PART 6

MACROECONOMIC POLICY IN THE OPEN ECONOMY 619

CHAPTER 25

Economic Policy in the Open Economy under Fixed Exchange Rates, 619

INTRODUCTION, 620 The Case of the Chinese Renminbi Yuan, 620 TITANS OF INTERNATIONAL ECONOMICS: ROBERT A. MUNDELL (BORN 1932), 621 TARGETS, INSTRUMENTS, AND ECONOMIC POLICY IN A TWO-INSTRUMENT, TWO-TARGET MODEL, 621 GENERAL EQUILIBRIUM IN THE OPEN ECONOMY: THE IS/LM/BP MODEL, 624 General Equilibrium in the Money Market: The LM Curve, 624 General Equilibrium in the Real Sector: The IS Curve, 628 Simultaneous Equilibrium in the Monetary and Real Sectors, 630 Equilibrium in the Balance of Payments: The BP Curve, 630 IN THE REAL WORLD: THE PRESENCE OF EXCHANGE CONTROLS IN THE CURRENT FINANCIAL SYSTEM, 635 Equilibrium in the Open Economy: The Simultaneous Use of the LM, IS, and BP Curves, 636 THE EFFECTS OF FISCAL POLICY UNDER FIXED EXCHANGE RATES, 639 THE EFFECTS OF MONETARY POLICY UNDER FIXED EXCHANGE RATES, 642 THE EFFECTS OF OFFICIAL CHANGES IN THE EXCHANGE RATE, 644 IN THE REAL WORLD: THE HISTORICAL RISE AND FALL OF A CURRENCY BOARD—THE CASE OF ARGENTINA, 646 SUMMARY, 648 Appendix, The Relationship between the Exchange Rate and Income in Equilibrium, 649

CHAPTER 26

Economic Policy in the Open Economy under Flexible Exchange Rates, 651

INTRODUCTION, 652
MOVEMENTS TO FLEXIBLE RATES, 652
THE EFFECTS OF FISCAL POLICY UNDER FLEXIBLE EXCHANGE RATES WITH DIFFERENT CAPITAL MOBILITY ASSUMPTIONS, 652
CONCEPT BOX 1: REAL AND FINANCIAL FACTORS THAT INFLUENCE THE BP CURVE, 654
THE EFFECTS OF MONETARY POLICY UNDER FLEXIBLE EXCHANGE RATES WITH DIFFERENT CAPITAL MOBILITY ASSUMPTIONS, 657
Policy Coordination under Flexible Exchange Rates, 659

THE EFFECTS OF EXOGENOUS SHOCKS IN THE IS/LM/BP MODEL WITH IMPERFECT MOBILITY OF CAPITAL, 661 IN THE REAL WORLD: COMMODITY PRICES AND U.S. REAL GDP, 1972-2014, 662 IN THE REAL WORLD: EUROPEAN INSTABILITY AND U.S. GDP. 666 IN THE REAL WORLD: POLICY FRICTIONS IN AN INTERDEPENDENT WORLD, 667 IN THE REAL WORLD: MACROECONOMIC POLICY COORDINATION: THE IMF, THE G-7/G-8, AND THE G-20, 669 SUMMARY, 670 Appendix, Policy Effects, Open-Economy Equilibrium, and the Exchange Rate under Flexible Rates, 671 CHAPTER 27 Prices and Output in the Open Economy: Aggregate Supply and Demand, 673 INTRODUCTION, 674 Crisis in Argentina, 674 AGGREGATE DEMAND AND SUPPLY IN THE CLOSED ECONOMY, 675 Aggregate Demand in the Closed Economy, 675 Aggregate Supply in the Closed Economy, 676 Equilibrium in the Closed Economy, 680 IN THE REAL WORLD: U.S. ACTUAL AND NATURAL INCOME AND UNEMPLOYMENT, 681 AGGREGATE DEMAND AND SUPPLY IN THE OPEN ECONOMY, 682 Aggregate Demand in the Open Economy under Fixed Rates, 683 Aggregate Demand in the Open Economy under Flexible Rates, 684 THE NATURE OF ECONOMIC ADJUSTMENT AND MACROECONOMIC POLICY IN THE OPEN-ECONOMY AGGREGATE SUPPLY AND Demand Framework, 685 Shifts in the Aggregate Demand Curve under Fixed and Flexible Rates, 685 The Effect of Monetary and Fiscal Policy on the Aggregate Demand Curve under Fixed and Flexible Rates, 686 Summary, 687 Monetary Policy in the Open Economy with Flexible Prices, 688 Currency Adjustments under Fixed Rates, 692 Fiscal Policy in the Open Economy with Flexible Prices, 692 Economic Policy and Supply Considerations, 693 IN THE REAL WORLD: ECONOMIC PROGRESS IN SUB-SAHARAN AFRICA, 695 EXTERNAL SHOCKS AND THE OPEN ECONOMY, 696 IN THE REAL WORLD: INFLATION AND UNEMPLOYMENT IN THE UNITED STATES, 1970-2014, 697 SUMMARY, 700

PART 7

ISSUES IN WORLD MONETARY ARRANGEMENTS 701

CHAPTER 28 **Fixed or Flexible Exchange Rates? 701** INTRODUCTION, 702 Exchange Rate Experiences, 702 CENTRAL ISSUES IN THE FIXED-FLEXIBLE EXCHANGE RATE DEBATE, 702 Do Fixed or Flexible Exchange Rates Provide for Greater "Discipline" on the Part of Policymakers? 702 Would Fixed or Flexible Exchange Rates Provide for Greater Growth in International Trade and Investment? 704 IN THE REAL WORLD: EXCHANGE RISK AND INTERNATIONAL TRADE, 705 Would Fixed or Flexible Exchange Rates Provide for Greater Efficiency in Resource Allocation? 706 Is Macroeconomic Policy More Effective in Influencing National Income under Fixed or Flexible Exchange Rates? 707 Will Destabilizing Speculation in Exchange Markets Be Greater under Fixed or Flexible Exchange Rates? 709 IN THE REAL WORLD: RESERVE HOLDINGS UNDER FIXED AND FLEXIBLE EXCHANGE RATES, 709 TITANS OF INTERNATIONAL ECONOMICS: MILTON FRIEDMAN (1912-2006), 713 Will Countries Be Better Protected from External Shocks under a Fixed or a Flexible Exchange Rate System? 714 IN THE REAL WORLD: "INSULATION" WITH FLEXIBLE RATES—THE CASE OF JAPAN, 715 CURRENCY BOARDS, 716 Advantages of a Currency Board, 716 IN THE REAL WORLD: CURRENCY BOARDS IN ESTONIA AND LITHUANIA, 717 Disadvantages of a Currency Board, 718 **OPTIMUM CURRENCY AREAS, 719** IN THE REAL WORLD: THE EASTERN CARIBBEAN CURRENCY UNION AND OTHER MONETARY UNIONS, 721 HYBRID SYSTEMS COMBINING FIXED AND FLEXIBLE EXCHANGE RATES, 722 Wider Bands, 722 Crawling Pegs, 723 Managed Floating, 724 IN THE REAL WORLD: COLOMBIA'S EXPERIENCE WITH A CRAWLING PEG, 725 SUMMARY, 726

CHAPTER 29

The International Monetary System: Past, Present, and Future, 728

INTRODUCTION, 729 Monetary System Uncertainties, 729 IN THE REAL WORLD: CONFIDENCE IN EXCHANGE RATES UNDER THE GOLD STANDARD, 730 IN THE REAL WORLD: FLEXIBLE EXCHANGE RATES IN POST-WORLD WAR I EUROPE: THE UNITED KINGDOM, FRANCE, AND NORWAY, 731 THE BRETTON WOODS SYSTEM, 732 The Goals of the IMF, 733 The Bretton Woods System in Retrospect, 736 GRADUAL EVOLUTION OF A NEW INTERNATIONAL MONETARY System, 737 Early Disruptions, 737 Special Drawing Rights, 737 The Breaking of the Gold–Dollar Link and the Smithsonian Agreement, 738 The Jamaica Accords, 739 The European Monetary System, 740 Exchange Rate Fluctuations in Other Currencies in the 1990s and 2000s, 743 CURRENT EXCHANGE RATE ARRANGEMENTS, 745 EXPERIENCE UNDER THE CURRENT INTERNATIONAL MONETARY System, 746

The Global Financial Crisis and the 2007-2009 Recession, 751 SUGGESTIONS FOR REFORM OF THE INTERNATIONAL MONETARY System, 753 A Return to the Gold Standard, 753 A World Central Bank, 754 CONCEPT BOX 1: A WORLD CENTRAL BANK WITHIN A THREE-CURRENCY MONETARY UNION, 755 The Target Zone Proposal, 756 Controls on Capital Flows, 758 Greater Stability and Coordination of Macroeconomic Policies across Countries, 759 IN THE REAL WORLD: THE EMERGENCE OF A NEW CURRENCY: BITCOIN, 760 International Monetary Arrangements and the Emerging/Developing Countries, 762 SUMMARY, 763

References for Further Reading, 765 Index, 784

CHAPTER

THE WORLD OF INTERNATIONAL ECONOMICS

INTRODUCTION

Welcome to the study of international economics. No doubt you have become increasingly aware of the importance of international transactions in daily economic life. When people say that "the world is getting smaller every day," they are referring not only to the increased speed and ease of transportation and communications but also to the increased use of international markets to buy and sell goods, services, and financial assets. This is not a new phenomenon, of course: in ancient times international trade was important for the Egyptians, the Greeks, the Romans, the Phoenicians, and later for Spain, Portugal, Holland, and Britain. It can be said that all the great nations of the past that were influential world leaders were also important world traders. Nevertheless, the importance of international trade and finance to the economic health and overall standard of living of a country has never been as clear as it is today.

Signs of these international transactions are all around us. The clothes we wear come from production sources all over the world: the United States to the Pacific Rim to Europe to Central and South America. The automobiles

we drive are produced not only in the United States but also in Canada, Mexico, Japan, Germany, France, Italy, England, Sweden, and other countries. The same can be said for the food we eat, the shoes we wear, the appliances we use, and the many different services we consume. In addition, in the United States, when you call an 800 number about a product or service, you may be talking to someone in India. Further, products manufactured in the United States often use important parts produced in other countries. At the same time, many U.S. imports are manufactured with important U.S.-made components.

This increased internationalization of economic life is made even more complicated by foreign-owned assets. More and more companies in many countries are owned partially or totally by foreigners. In the 1990s, foreigners began to purchase U.S. government bonds and corporate stocks in record numbers, partly fueling the stock market boom of those years. The overall heightened presence of foreign goods, foreign producers, and foreign-owned assets causes many to question the impact and desirability of international transactions. This questioning became more intense with the onset of the 2007–2008 global financial crisis and accompanying recession.

These recent events draw attention to the changes in the world economic structure that have been taking place in the years following World War II (WWII) when the United States was the dominant world economic power. The nature of that dominance has, however, been undergoing considerable change in terms of world production structure and importance in global GDP and trade. The emergence of Japan, the European Union (EU), and, notably, China on the world scene has been significant. Indeed, in terms of relative importance in global GDP and trade, the United States and China are roughly equivalent. The nature of economic activity both in terms of location and structure has also been changing as world production has become more integrated through global supply chains and the evolution of the world international monetary system. Even though the United States has diminished in relative importance in global services and in areas such as technology development, and electronic development/processing. In addition, every country is affected by actions of the U.S. Federal Reserve as the U.S. dollar remains the world's key currency.

You will be studying one of the oldest branches of economics. People have been concerned about the goods and services crossing their borders for as long as nation-states or city-states have existed. Some of the earliest economic data relate to international trade, and early economic thinking often centered on the implications of international trade for the well-being of a politically defined area. Although similar to regional economics in many respects, international economics has traditionally been treated as a special branch of the discipline. This is not terribly surprising when one considers that economic transactions between politically distinct areas are often associated with many differences that influence the nature of exchanges between them rather than transactions within them. For example, the degree of factor mobility between countries often differs from that within countries. Countries can have different forms of government, different currencies, different types of economic systems, different resource endowments, different cultures, different institutions, and different arrays of products.

The study of international economics, like all branches of economics, concerns decision making with respect to the use of scarce resources to meet desired economic objectives. It examines how international transactions influence such things as social welfare, income distribution, employment, growth, and price stability, and the possible ways public policy can affect the outcomes. In the study of international trade, we ask, for example: What determines the basis for trade? What are the effects of trade? What determines the value and the volume of trade? What factors impede trade flows? What is the impact of public policy that attempts to alter the pattern of trade? In the study of international monetary economics we address questions such as: What is meant by a country's balance of payments? How are

exchange rates determined? How does trade affect the economy at the macro level? Why does financial capital flow rapidly and sizably across country borders? Should several countries adopt a common currency? How do international transactions affect the use of monetary and fiscal policy to pursue domestic targets? How do economic developments in a country get transmitted to other countries? This chapter provides an overview of the subjects and issues of international economics that will be discussed throughout the rest of this text.

THE NATURE OF MERCHANDISE TRADE

Before delving further into the subject matter of international economics, however, it is useful to take a brief look at some of the characteristics of world trade today. The value of world merchandise exports was \$18.9 trillion in 2014, a figure that is dramatic when one realizes that the value of goods exported worldwide was less than \$2 trillion in 1985. Throughout the past four decades, international trade volume has, on average, outgrown production (see Table 1), illustrating how countries are becoming more interdependent. With the worldwide recession, slow recovery, and uncertainties of recent years, trade growth has been variable: It grew 2.5 percent in 2014, 3.0 percent in 2013, 2.5 percent in 2012, 5.0 percent in 2011, 13.8 percent in 2010, and a *negative* 12.0 percent in 2009.¹

The Geographical Composition of Trade

In terms of major economic areas, the industrialized countries dominate world trade. Details of trade on a regional basis are provided in Table 2. The relative importance of Europe, North America, and Asia is evident, as they account for more than 83 percent of trade. Asia has become increasingly important in developing countries' imports and exports.

To obtain an idea of the geographical structure of trade, look at Table 3, which provides information on the destination of merchandise exports from several regions for 2014. The first row, for example, indicates that 50.2 percent of the exports of countries of North America went to other North American countries, 8.6 percent of North American exports

TABLE 1 Growth in Volume of World Goods Production and Trade, 1963–2014 (average annual percentage change in volume)

| | 1963-1973 | 1970–1979 | 1980–1985 | 1985–1990 | 1990–1998 | 1995–2000 | 2000–2006 | 2005–2010 | 2010–2014 | 2014 |
|-----------------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|------|
| Production | | | | | | | | | | |
| All commodities | 6.0% | 4.0% | 1.7% | 3.0% | 2.0% | 4.0% | 2.5% | 2.0% | 2.5% | 2.0% |
| Agriculture | 2.5 | 2.0 | 2.9 | 1.9 | 2.0 | 2.5 | 2.0 | 2.0 | 2.5 | 1.5 |
| Mining | 5.5 | 2.5 | -2.7 | 3.0 | 2.0 | 2.0 | 1.5 | 0.5 | 1.5 | 2.5 |
| Manufacturing | 7.5 | 4.5 | 2.3 | 3.2 | 2.0 | 4.0 | 3.0 | 2.5 | 2.5 | 2.5 |
| Exports | | | | | | | | | | |
| All commodities | 9.0% | 5.0% | 2.1% | 5.8% | 6.5% | 7.0% | 5.5% | 3.5% | 3.5% | 2.5% |
| Agriculture | 14.0 | 4.5 | 1.0 | 2.2 | 4.0 | 3.5 | 4.0 | 3.5 | 3.5 | 2.5 |
| Mining | 7.5 | 1.5 | -2.7 | 4.8 | 5.5 | 4.0 | 3.0 | 1.5 | 1.5 | 1.0 |
| Manufacturing | 11.5 | 7.0 | 4.5 | 7.0 | 7.0 | 8.0 | 6.0 | 4.0 | 4.0 | 4.0 |
| | | | | | | | | | | |

Sources: General Agreement on Tariffs and Trade, International Trade 1985–86 (Geneva: GATT, 1986), p. 13; GATT, International Trade 1988–89, I (Geneva: GATT, 1989), p. 8; GATT, International Trade 1993: Statistics (Geneva: GATT, 1993), p. 2; GATT, International Trade 1994: Trends and Statistics (Geneva: GATT, 1994), p. 2; World Trade Organization, Annual Report 1999: International Trade Statistics (Geneva: WTO, 1999), p. 1; WTO, International Trade Statistics 2003 (Geneva: WTO, 2003), p. 19; WTO, International Trade Statistics 2007 (Geneva: WTO, 2007), p. 7; WTO, International Trade Statistics 2011 (Geneva: WTO, 2011), p. 19; and WTO, World Trade Report 2015, p. 39, all obtained from www.wto.org.

¹World Trade Organization, *International Trade Statistics 2015*, Table 1.1, and World Trade Organization, Press Release 658, April 12, 2012, "Trade Growth to Slow in 2012 after Strong Deceleration in 2011," both obtained from www.wto.org.

| | Exports | | Imports | | |
|--|-----------------------------------|-----------|------------------------------------|-----------|--|
| | (billions of dollars, f.o.b.*) | Share (%) | (billions of dollars, c.i.f. *) | Share (%) | |
| North America [†] | \$ 2,493 | 13.2% | \$ 3,300 | 17.4% | |
| South and Central America | 695 | 3.7 | 739 | 3.9 | |
| Europe | 6,739 | 35.6 | 6,722 | 35.3 | |
| (European Union) [‡] | (6,162) | (32.6) | (6,133) | (32.2) | |
| Commonwealth of Independent States (CIS) [§] | 735 | 3.9 | 506 | 2.7 | |
| Africa | 555 | 2.9 | 642 | 3.4 | |
| Middle East | 1,288 | 6.8 | 784 | 4.1 | |
| Asia | 6,426 | 33.9 | 6,325 | 33.3 | |
| World | \$18,930 | 100.0% | \$19,018 | 100.0% | |

TABLE 2 Merchandise Exports and Imports by Region, 2014 (billions of dollars and percentage of world totals)

Note: Components may not sum to totals because of rounding.

*Exports are recorded f.o.b. (free on board) and imports are recorded c.i.f. (cost, insurance, and freight).

[†]Including Mexico.

[‡]Austria, Belgium, Bulgaria, Croatia, Cyprus, Czech Republic, Denmark, Estonia, Finland, France, Germany, Greece, Hungary, Ireland, Italy, Latvia, Lithuania, Luxembourg, Malta, Netherlands, Poland, Portugal, Romania, Slovak Republic, Slovenia, Spain, Sweden, and United Kingdom.

[§]Armenia, Azerbaijan, Belarus, Georgia, Kazakhstan, Kyrgyz Republic, Moldova, Russian Federation, Tajikistan, Turkmenistan, Ukraine, and Uzbekistan.

Source: World Trade Organization, World Trade Report 2015, pp. 24, 26, obtained from www.wto.org.

TABLE 3 Regional Structure of World Merchandise Exports, 2014 (percentage of each origin area's exports going to each destination area)

| | | Destination | | | | | | |
|---------------------------|------------------|------------------------------|--------|------|--------|-------------|-------|--------|
| Origin | North America | South and Central America | Europe | CIS | Africa | Middle East | Asia | World |
| North America | 50.2% | 8.6% | 15.2% | 0.7% | 1.7% | 3.2% | 20.2% | 100.0% |
| South and Central America | 24.8 | 25.8 | 16.4 | 1.4 | 2.5 | 2.4 | 24.5 | 100.0 |
| Europe | 7.9 | 1.7 | 68.5 | 3.2 | 3.3 | 3.4 | 10.8 | 100.0 |
| CIS | 3.9 | 0.9 | 52.4 | 17.8 | 2.1 | 3.1 | 18.2 | 100.0 |
| Africa | 7.0 | 5.1 | 36.2 | 0.4 | 17.7 | 3.3 | 27.3 | 100.0 |
| Middle East | 7.7 | 0.8 | 11.5 | 0.5 | 2.8 | 8.8 | 53.9 | 100.0 |
| Asia | 18.0 | 3.1 | 15.2 | 2.1 | 3.5 | 5.1 | 52.3 | 100.0 |
| World | 17.3 | 4.0 | 36.7 | 2.8 | 3.5 | 4.2 | 29.7 | 100.0 |

Note: Destination percentages for any given origin area do not sum to 100.0% because of rounding and/or incomplete specification.

Source: World Trade Organization, International Trade Statistics 2015, p. 41, obtained from www.wto.org.

went to South and Central America, and so forth. From this table it is clear that the major markets for all regions' exports are in North America, Europe, and Asia. This is true for these three areas themselves, especially for Europe, which sends 68.5 percent of its exports to itself. In addition, the table makes it evident that the countries in Africa and the Middle East do not trade heavily among themselves.

At the individual country level (see Table 4), the relative importance of Europe, North America, and Asia in 2014 is again quite evident. The largest country exporter is China

| Exports | | | | Imports | | | | |
|--------------------------|----------|--------|----------------------|----------|--------|--|--|--|
| Country | Value | Share | Country | Value | Share | | | |
| 1. China | \$ 2,342 | 12.4% | United States | \$ 2,413 | 12.7% | | | |
| 2. United States | 1,621 | 8.6 | China | 1,959 | 10.3 | | | |
| 3. Germany | 1,508 | 8.0 | Germany | 1,216 | 6.4 | | | |
| 4. Japan | 684 | 3.6 | Japan | 822 | 4.3 | | | |
| 5. Netherlands | 672 | 3.6 | United Kingdom | 684 | 3.6 | | | |
| 6. France | 583 | 3.1 | France | 678 | 3.6 | | | |
| 7. Republic of Korea | 573 | 3.0 | Hong Kong (China) | 601 | 3.2 | | | |
| 8. Italy | 529 | 2.8 | Netherlands | 588 | 3.1 | | | |
| 9. Hong Kong (China) | 524 | 2.8 | Republic of Korea | 526 | 2.8 | | | |
| 10. United Kingdom | 506 | 2.7 | Canada* | 475 | 2.5 | | | |
| 11. Russian Federation | 498 | 2.6 | Italy | 472 | 2.5 | | | |
| 12. Canada | 475 | 2.5 | India | 463 | 2.4 | | | |
| 13. Belgium | 471 | 2.5 | Belgium | 452 | 2.4 | | | |
| 14. Singapore | 410 | 2.2 | Mexico | 412 | 2.2 | | | |
| 15. Mexico | 398 | 2.1 | Singapore | 366 | 1.9 | | | |
| 16. United Arab Emirates | 360 | 1.9 | Spain | 358 | 1.9 | | | |
| 17. Saudi Arabia | 354 | 1.9 | Russian Federation* | 308 | 1.6 | | | |
| 18. Spain | 325 | 1.7 | Taiwan | 274 | 1.4 | | | |
| 19. India | 322 | 1.7 | United Arab Emirates | 262 | 1.4 | | | |
| 20. Taiwan | 314 | 1.7 | Turkey | 242 | 1.3 | | | |
| 21. Australia | 241 | 1.3 | Brazil | 239 | 1.3 | | | |
| 22. Switzerland | 239 | 1.3 | Australia | 237 | 1.2 | | | |
| 23. Malaysia | 234 | 1.2 | Thailand | 228 | 1.2 | | | |
| 24. Thailand | 228 | 1.2 | Poland | 220 | 1.2 | | | |
| 25. Brazil | 225 | 1.2 | Malaysia | 209 | 1.1 | | | |
| 26. Poland | 217 | 1.1 | Switzerland | 203 | 1.1 | | | |
| 27. Austria | 178 | 0.9 | Austria | 182 | 1.0 | | | |
| 28. Indonesia | 176 | 0.9 | Indonesia | 178 | 0.9 | | | |
| 29. Czech Republic | 174 | 0.9 | Saudi Arabia | 163 | 0.9 | | | |
| 30. Sweden | 164 | 0.9 | Sweden | 163 | 0.9 | | | |
| 30 countries | \$15,542 | 82.1% | | \$15,592 | 82.0% | | | |
| World | \$18,930 | 100.0% | | \$19,018 | 100.0% | | | |

| TABLE 4 | Leading Merchandise Exporters and Importers, 2014 (billions of dollars and percentage share of world totals) |
|---------|--|
|---------|--|

Note: Components do not sum to totals because of rounding.

*Imports valued f.o.b.

Source: World Trade Organization, World Trade Report 2015, p. 26, obtained from www.wto.org.

(which displaced Germany in 2009, which in turn had displaced the United States in 2003). The six largest traders (exports plus imports) are China, the United States, Germany, Japan, France, and the Netherlands, and they account for nearly 40 percent of world trade. Also noteworthy has been the spectacular growth in the trade of Hong Kong, the Republic of Korea (South Korea), Taiwan, and Singapore. Finally, the 10 largest trading countries account for over 50 percent of world trade. World trade thus tends to be concentrated among relatively few major traders, with the remaining approximately 200 countries accounting for slightly less than 50 percent.

The Commodity Composition of Trade

Turning to the 2014 commodity composition of world trade (Table 5), manufactures account for 66.5 percent of trade, with the remaining amount consisting of primary products. Among primary goods, trade in fuels is the largest (16.7 percent), followed by food products (8.1 percent). Trade in raw materials, ores and other minerals, and nonferrous metals accounts for 5.4 percent. In the manufacturing category, machinery and transport equipment account for 33.2 percent of world trade. Office and telecom equipment and automotive/transport products are major subcategories, accounting for 9.7 percent and 11.4 percent of exports, respectively. Other important categories of manufactures include trade in chemicals (11.1 percent) and in textiles and clothing (4.3 percent).

What is especially notable is the current importance of trade in manufactures and the declining importance of primary products. Comparison of the last column of Table 5 with the next-to-last column illustrates the relatively sluggish growth of primary products in world trade compared with the growth in manufactured goods. For example, food products accounted for 11.0 percent of world exports in 1980 but only 8.1 percent in 2014; fuels, which constituted 23.0 percent in 1980, fell in importance to 16.7 percent in 2014; and the share of primary products in total dropped from 42.4 percent in 1980 to slightly over 30 percent in 2014. These developments are of particular relevance to many developing countries, whose trade has traditionally been concentrated in primary goods. Specialization in commodity groups that are growing relatively more slowly makes it difficult for them to obtain the gains from growth in world trade accruing to countries exporting manufactured products. The demand for primary products not only tends to be less responsive to income growth but also is more likely to demonstrate greater price fluctuations.

| Product Category | Value in 2014 (\$ billion) | Share in 2014 | Share in 1980 |
|-----------------------------------|----------------------------|---------------|---------------|
| Agricultural products | \$ 1,765 | 9.6% | 14.7% |
| Food | 1,486 | 8.1 | 11.0 |
| Raw materials | 279 | 1.5 | 3.7 |
| Fuels and mining products | 3,789 | 20.6 | 27.7 |
| Ores and other minerals | 366 | 2.0 | 2.1 |
| Fuels | 3,068 | 16.7 | 23.0 |
| Nonferrous metals | 354 | 1.9 | 2.5 |
| Manufactures | 12,243 | 66.5 | 53.9 |
| Iron and steel | 472 | 2.6 | 3.8 |
| Chemicals | 2,054 | 11.1 | 7.0 |
| Other semimanufactures | 1,185 | 6.4 | 6.7 |
| Machinery and transport equipment | 6,112 | 33.2 | 25.8 |
| Office and telecom equipment | 1,794 | (9.7) | (4.2) |
| Automotive products and other | | | |
| transport equipment | 2,107 | (11.4) | (6.5) |
| Other machinery | 2,211 | (12.0) | (15.2) |
| Textiles | 314 | 1.7 | 2.7 |
| Clothing | 483 | 2.6 | 2.0 |
| Other manufactures | 1,623 | 8.8 | 5.8 |
| Total | \$18,422 | 100.0% | 100.0% |

TABLE 5 Commodity Composition of World Exports, 2014 and 1980

Note: Components may not sum to category totals because of rounding. The three aggregate categories do not sum to \$18,422 and 100.0% because of incomplete specification of products.

Sources: World Trade Organization, International Trade 1995: Trends and Statistics (Geneva: WTO, 1995), p. 77; WTO, World Trade Report 2015, p. 24, and Appendix Table A12, obtained from www.wto.org.

| | Exports | to | Imports | Imports from | |
|-------------------------|-------------|--------|-------------|----------------|--|
| Region or Country | Value | Share | Value | Share 20.9% | |
| Europe | \$ 337,024 | 20.6% | \$ 495,893 | | |
| European Union | 279,127 | 17.1 | 422,580 | 15.7 | |
| Belgium | 34,752 | 2.1 | 21,160 | 0.9 | |
| France | 31,684 | 1.9 | 47,606 | 2.0 | |
| Germany | 49,637 | 3.0 | 124,174 | 5.2 | |
| Ireland | 7,907 | 0.5 | 34,088 | 1.4 | |
| Italy | 17,123 | 1.0 | 42,451 | 1.8 | |
| Netherlands | 43,195 | 2.6 | 21,363 | 0.9 | |
| Switzerland | 22,552 | 1.4 | 31,370 | 1.3 | |
| United Kingdom | 54,547 | 3.3 | 55,439 | 2.3 | |
| Non-European Union | 57,897 | 3.5 | 73,313 | 3.1 | |
| Canada | 313,510 | 19.2 | 354,354 | 14.9 | |
| Latin America and Other | | | | | |
| Western Hemisphere | 425,401 | 26.1 | 452,876 | 19.1 | |
| Brazil | 42,412 | 2.6 | 30,102 | 1.3 | |
| Mexico | 240,721 | 14.7 | 301,403 | 12.7 | |
| Venezuela | 11,129 | 0.7 | 30,339 | 1.3 | |
| Asia and Pacific | 445,305 | 27.3 | 931,750 | 39.2 | |
| China | 124,747 | 7.6 | 467,940 | 19.7 | |
| Hong Kong (China) | 41,997 | 2.6 | 6,345 | 2.6 | |
| India | 22,523 | 1.4 | 45,412 | 1.9 | |
| Japan | 68,014 | 4.2 | 136,680 | 5.8 | |
| Republic of Korea | 46,114 | 2.8 | 69,846 | 2.9 | |
| Singapore | 30,063 | 1.8 | 16,456 | 0.7 | |
| Taiwan | 27,135 | 1.7 | 40,700 | 1.7 | |
| Middle East | 73,434 | 4.5 | 104,350 | 4.4 | |
| Israel | 15,094 | 0.9 | 23,199 | 0.9 | |
| Saudi Arabia | 17,866 | 1.1 | 47,125 | 1.8 | |
| Africa | 37,670 | 2.3 | 34,879 | 1.5 | |
| Nigeria | 5,982 | 0.4 | 3,896 | 0.2 | |
| (Members of OPEC) | (81,372) | (4.9) | (133,198) | (5.6) | |
| Total | \$1,632,639 | 100.0% | \$2,374,101 | 100.0% | |

TABLE 6U.S. Merchandise Trade by Area and Country, 2014 (millions of dollars
and percentage shares)

Notes: (a) Components may not sum to totals because of rounding; and (b) data are preliminary.

Source: U.S. Department of Commerce, Bureau of Economic Analysis, Table 2.3 of Press Release, September 17, 2015, obtained from www.bea.gov.

U.S. International Trade

To complete our discussion of the current nature of merchandise trade, we take a closer look at the geographic and commodity characteristics of the 2014 U.S. international trade (see Tables 6 and 7). Geographically, Canada is the most important trading partner for the United States, both in exports and imports. The North American Free Trade Agreement (NAFTA) partners (Canada and Mexico) are the largest multicountry unit, followed by the EU. The second-largest individual trading partner country of the United States, behind Canada, is China, followed by Mexico, Japan, Germany, Republic of Korea, the United Kingdom, France, Brazil, and Taiwan. Of note is the fact that a major portion (63.7 percent) of the trade deficit of the United States in 2014 could be traced to China, Japan, and Mexico.

| TABLE 7 Composition of U.S. Trade, 2014 (billions of dollars and percentage sha | TABLE 7 | Composition of U.S. Tra | le, 2014 (billions of dollars | and percentage shares) |
|---|---------|--------------------------------|-------------------------------|------------------------|
|---|---------|--------------------------------|-------------------------------|------------------------|

| | V-h-af E | Ch | Value of Y | CL |
|--|------------------|-------------------|---------------------|-----------|
| | Value of Exports | Share (%) | Value of Imports | Share (%) |
| Foods, feeds, and beverages | 143.8 | 8.8 | 126.7 | 5.4 |
| Coffee, cocoa, and sugar | | | 8.2 | 0.3 |
| Fish and shellfish | 6.0 | 0.4 | 20.2 | 0.9 |
| Grains and preparations | 32.8 | 2.0 | _ | |
| Meat products and poultry | 20.1 | 1.2 | 12.1 | 0.5 |
| Soybeans | 25.1 | 1.5 | _ | _ |
| Vegetables, fruits, nuts, and preparations | 25.2 | 1.5 | 28.2 | 1.2 |
| Industrial supplies and materials | 500.0 | 30.6 | 672.6 | 28.5 |
| Building materials, except metals | 15.5 | 0.9 | 28.2 | 1.2 |
| Chemicals, excluding medicinals | 119.3 | 7.3 | 80.7 | 3.4 |
| Energy products | 182.5 | 11.2 | 375.0 | 15.9 |
| Metals and nonmetallic products | 89.7 | 5.5 | 114.6 | 1.8 |
| Iron and steel products | 19.0 | 1.2 | 46.2 | 2.0 |
| Nonferrous metals | 33.3 | 2.0 | 37.6 | 1.6 |
| Paper and paper base stocks | 22.5 | 1.4 | 12.6 | 0.5 |
| Textile supplies and related materials | 15.2 | 0.9 | 14.8 | 0.6 |
| Capital goods, except automotive | 551.3 | 33.0 | 595.7 | 25.3 |
| Civilian aircraft, engines, and parts | 113.1 | 6.9 | 53.2 | 2.3 |
| Machinery and equipment, except consumer-type | 431.5 | 26.4 | 535.5 | 22.7 |
| Computers, peripherals, and parts | 31.9 | 2.0 | 121.7 | 5.2 |
| Electric generating machinery, electric apparatus, and parts | 57.3 | 3.5 | 71.3 | 3.0 |
| Industrial engines, pumps, and compressors | 29.7 | 1.8 | 24.8 | 1.0 |
| Machine tools and metalworking machinery | 7.7 | 0.5 | 11.4 | 0.5 |
| Measuring, testing, and control instruments | 25.2 | 1.5 | 20.0 | 0.8 |
| Oil drilling, mining, and construction machinery | 29.6 | 1.8 | 24.1 | 1.0 |
| Scientific, hospital, and medical equipment and parts | 46.2 | 2.8 | 40.2 | 1.7 |
| Semiconductors | 43.7 | 2.7 | 44.0 | 1.9 |
| Telecommunications equipment | 40.7 | 2.5 | 58.7 | 2.5 |
| Automotive vehicles, parts, and engines | 159.7 | 9.8 | 328.5 | 13.9 |
| (to/from Canada) | (60.0) | (3.7) | (63.2) | (2.7) |
| Passenger cars, new and used | 60.6 | 3.8 | 153.6 | 6.5 |
| Trucks, buses, and special purpose vehicles | 19.9 | 1.2 | 32.5 | 1.4 |
| Engines and engine parts | 18.4 | 1.1 | 29.4 | 1.2 |
| Other parts and accessories | 60.9 | 3.8 | 113.1 | 4.8 |
| Consumer goods (nonfood), except automotive | 198.3 | 12.1 | 559.4 | 23.7 |
| Durable goods | 110.8 | 6.8 | 301.3 | 12.8 |
| Household and kitchen appliances and other household goods | 40.2 | 2.5 | 164.8 | 7.0 |
| Radio and stereo equipment, including records, tapes, and disks | 4.8 | 0.3 | 9.8 | 0.4 |
| Televisions, video receivers, and other video equipment | 4.6 | 0.3 | 28.2 | 1.2 |
| Toys and sporting goods, including bicycles | 10.4 | 0.6 | 36.9 | 1.6 |
| Nondurable goods | 87.5 | 5.4 | 258.1 | 10.9 |
| Apparel, footwear, and household goods | 87.5 11.3 | 5.4 0.7 | 135.9 | 5.8 |
| | 51.0 | | | |
| Medical, dental, and pharmaceutical preparations | 51.0 56.6 | 3.1 3.5 | 91.9 75.7 | 3.9 |
| Goods, not elsewhere classified (including U.S. import goods returned) | | | | 3.2 |
| Total | \$1,609.7 | 100.0% | \$2,358.7 | 100.0% |

Notes: (a) Major category figures may not sum to totals because of rounding; (b) — = not available or negligible; and (c) data are preliminary.

Source: U.S. Department of Commerce, Bureau of Economic Analysis, Table 2.1 of Press Release, September 17, 2015, obtained from www.bea.gov.

Turning to the commodity composition of U.S. trade (Table 7), agricultural products (foods, feeds, and beverages) are an important source of exports. The capital goods category is the largest single export category and is dominated by nonelectric machinery. Industrial supplies, importantly consisting of chemicals and metal/nonmetallic products, is also an important export category for the United States. If energy products are excluded, this category would show a small trade surplus. Sizeable net imports occur in consumer goods, autos, and energy products. The largest import category is industrial supplies and materials, followed by capital goods (except automotive) and consumer goods imports.

WORLD TRADE IN SERVICES

The discussion of world trade has to this point focused on merchandise trade and has ignored the rapidly growing exports of services, estimated to be almost \$5 trillion in 2014 (one-fifth of total exports in goods and services). The rising importance of services in trade should not be unexpected since the service category now accounts for the largest share of income and employment in many industrial countries including the United States. More specifically, in recent years services accounted for 79 percent of gross domestic product (GDP) in France, 69 percent in Germany, 78 percent in the United States, 78 percent in the United Kingdom, and 73 percent in Japan.² In this context, services generally include the following categories in the International Standard Industrial Classification (ISIC) system: wholesale and retail trade, restaurants and hotels, transport, storage, communications, financial services, insurance, real estate, business services, personal services, community services, social services, and government services.

International trade in services broadly consists of commercial services, investment income, and government services, with the first two categories accounting for the bulk of services. Discussions of trade in "services" generally refer to trade in commercial services. During the 1970s this category grew more slowly in value than did merchandise trade. However, since that time, exports of commercial services have outgrown merchandise exports, and the relative importance of commercial services is roughly the same today as it was in the early 1970s. A word of caution is in order, however: the nature of trade in "services" is such that it is extremely difficult to obtain accurate estimates of the value of these transactions. This results from the fact that there is no agreed definition of what constitutes a traded service, and the ways in which these transactions are measured are less precise than is the case for merchandise trade. Estimates are obtained by examining foreign exchange records and/or through surveys of establishments. Because many service transactions are not observable (hence, they are sometimes referred to as the "invisibles" in international trade), the usual customs records or data are not available for valuing these transactions. Thus, it is likely that the value of trade in commercial services is underestimated. However, there may also be instances when firms may choose to overvalue trade in services, and reported figures must be viewed with some caution.

In terms of the geographical nature of trade in services, this trade is also concentrated among the industrial countries (see Table 8). The principal world traders in merchandise are generally also the principal traders in services. It is notable that both exports and imports of services are important for industrializing economies such as Thailand, Taiwan, and India.

The nature of trade in services is such that until the 1980s they were virtually ignored in trade negotiations and trade agreements. However, because of their increasing importance, there has been a growing concern for the need to establish some general guidelines for international transactions in services. Consequently, discussions regarding the nature of the service trade and various country restrictions that may influence it were included in the

²World Bank, World Development Indicators 2015, Table 4.2, obtained from http://wdi.worldbank.org/table/4.2.